

## Law Enforcement Guidance

**Prepared** for

Drug Control and Access to Medications (DCAM) Consortium

October 2010

APMG

108/1 Erskineville Rd Newtown 2042 Australia Telephone +61 2 9519 1039 Facsimile +61 2 9517 2039 admin@aidsprojects.com www.aidsprojects.com ACN 106 054 326

### Table of contents

1. Introduction	3
A. Why this guide?	3
B. Who is this guide for?	4
C. What principles should guide the role of law enforcement in facilitating access to medical opioids?	
D. Opioids for pain relief, heroin dependence and Medication-Assisted Treatment (MAT)	
2 Ensuring the safe manufacture, import, transport, and storage of medical opioids	10
A. Summary of Drug Convention requirements	10
B. Control to ensure availability: experience from selected countries	
Guidance points	14
3. Diversion control: not hindering access by prescribers and dispensers, and by end-users, to	
medical opioids	15
A. Diversion control: the unintended impact	16
B. Important considerations towards developing diversion control regulations, policies and practices	
Guidance points	
MAT specific guidance	
4. Law enforcement assisting access by end-users to medical opioids	22
A. Working partnerships between law enforcement and health agencies (including hospitals / clinics providing	
pain relief and MAT providers)	
B. Pre-arrest referral of people who use drugs illicitly to health and welfare groups	
C. Post arrest referral of non-violent drug offenders to drug treatment	
Guidance points	
5. Assisting ongoing treatment for people in, and released from, custodial settings	30
A The need for treatment continuity	
B Common concerns among law enforcement and corrections personnel related to MAT in prisons	
C Examples of good practice from Iran, India, Poland, USA and Australia.	
Guidance points	
6. Conclusion	36
7. DCAM website	38
8. Annexes	39
References	

### ACKNOWLEDGMENTS

This training guide was developed by AIDS Projects Management Group. It was written by Ruth Birgin, Paul Hardacre and Dave Burrows.

Please cite as:

Birgin R, Hardacre P and Burrows D. Law Enforcement Guidance: Prepared for Drug Control and Access to Medications Consortium APMG. 2010. Sydney

### 1. Introduction

Millions of people suffer from severe untreated pain caused by cancer, HIV, accidents, surgery, and chronic conditions. Less than 5% of the estimated 13 million heroin dependent people have

access to effective, Medication-Assisted Treatment (or MAT<sup>1</sup> – *See section 1.D for a detailed definition of MAT*). In most jurisdictions, the opioid medicines required to effectively manage and treat pain, and MAT for heroin dependence in particular, are unavailable or inaccessible to the majority of those in need. International treaties – namely, the 1961 Single Convention on Narcotic Drugs, as amended by the 1972 Protocol; the 1971 Convention on Psychotropic Substances, and the 1988 Convention Against Illicit Traffic in Narcotic Drugs and Psychotropic Substances – oblige governments to provide access to opioid medicines for pain and heroin dependency. The recent United Nations Commission on Narcotic Drugs session also emphasized that drug control conventions seek a balance between ensuring the availability of opioids for medical and scientific purposes and preventing their diversion, and encourages Member States to consider ways to leverage existing health and development programmes in countries without adequate availability of opioids for medical and scientific purposes.<sup>1</sup>

At all levels it is increasingly recognized that unnecessarily restrictive drug control laws, regulations and practices, training shortfalls, weak healthcare systems, and apprehension among healthcare workers regarding legal sanctions for legitimate opioid prescription have combined to perpetuate untreated pain and heroin dependency in many jurisdictions.

### A. Why this guide?

The role of law enforcement is key – not only as a component of international drug regulation and control systems, but as a facilitator of access to opioid medicines by those in need.

The World Health Organization (WHO) encourages the following practices for law enforcement in order to better support appropriate access to opioid medicines:

- ensuring that not interfering in appropriate access to medical opioids is an explicit goal
  of law enforcement practice, in balance with control of illegal use;
- that the policing of opioid prescription is conducted with well informed sensitivity for the diversity and importance of discretion in physician decision making;
- that activities that would deter people who inject drugs from seeking MAT and other forms of drug treatment and public health services are avoided;
- that MAT is available within the criminal justice system through diversion programmes or prison health services.<sup>2</sup>

Law enforcement personnel can and do hinder access to medical opioids and MAT, but equally can play a key role in reforming and streamlining legitimate access without compromising effective policing of drug diversion. A functioning medical opioids supply and delivery system requires law enforcement officers to differentiate between good medicine and drug trafficking, and to effectively identify and control key sources of diversion. Effective implementation also

<sup>&</sup>lt;sup>1</sup> Medication-Assisted Treatment, using long-acting drugs such as methadone or buprenorphine, is currently the most effective treatment available for heroin dependence

requires attitudinal changes among law enforcement officials to support reforms in practice. "Law enforcement agencies play an important role in promoting, or inhibiting, healthy behavior."<sup>3</sup>

Section 1 of this guide explains the context for a more balanced approach regarding the control of medical opioids, and provides definitions of key subject areas and explanations of important background concepts. Rationale, examples and models are described for later adaptation to specific national contexts. United Nations (UN) Convention requirements and practical guidance for the law enforcement role in medical opioid supply – from manufacture to storage – is drawn from the experience of several countries and described in Section 2. Section 3 explains the need for a more considered approach to diversion control and provides some central guidance. Section 4 provides a range of examples of good practice in law enforcement towards improving access to medical opioid access for those in need. Section 5 details ways of improving continuity in medical opioid access for those moving into, and out of, custodial settings. The guide also offers links to relevant supporting documents including those available via the Drug Control and Access to Medicines (DCAM) Consortium website, which centralizes relevant resources.

### B. Who is this guide for?

This guide is designed to provide instructive information to all levels of law enforcement, particularly narcotics enforcement at the local, state and national levels, corrective services and custodial officers, and drug regulators.

## C. What principles should guide the role of law enforcement in facilitating access to medical opioids?

The International Narcotics Control Board (INCB) recognizes that "the Single Convention on Narcotic Drugs of 1961 as amended by the 1972 Protocol establishes a dual drug control obligation for Governments: to ensure adequate availability of narcotic drugs, including opiates, for medical and scientific purposes, while at the same time preventing the illicit production of, trafficking in and use of such drugs."<sup>4</sup> In addition, the INCB and the WHO have endorsed the principle of balance for use in assessing national drug control policies.<sup>5</sup>

As part of a blueprint for reforming access to opioid medications, the principle of balance is emphasised. Balance means that the important responsibility to prevent illicit drug trafficking and abuse must not interfere in ensuring that therapeutic opioid medications are available and accessible to patients suffering from pain or drug dependence. This principle also draws attention to the importance of balancing associated law reform with a commitment to effective implementation within an expanded system of medical access which better ensures required flow of opioid medication and MAT to end users.<sup>6</sup>

Other guiding principles and approaches of particular relevance to MAT provision are articulated in the UN Drug Control Conventions, the Declaration on the Guiding Principles of Drug Demand Reduction and UN human rights and health promotion policies. These highlight that:

• protection of human rights is critical for preventing HIV as people are more vulnerable to infection when their economic, health, social, or cultural rights are not respected. Equally, a punitive approach, overly reliant on criminal justice measures, succeeds only

in driving underground those people most in need of prevention and care services;  $^{7\ 8\ 9\ 10}$ 

• flexible, easy-to-access MAT is critical to meeting the needs of heroin dependent people.<sup>15</sup>

In relation to Section 5 of this guide on pain management and MAT for prisoners, the current UN system position paper on HIV prevention, care, treatment and support in prison calls on countries to provide "adequate pain management and access to effective, appropriate and compassionate palliative care to a quality standard equivalent to that of the outside community."<sup>16</sup> WHO also recommends that patients of MAT taken into custody should be able to continue such treatment.<sup>17</sup> The guiding principle is that people moving into, and out of, custody should be provided with continuity of treatment with medical opioids and/or MAT.

### D. Opioids for pain relief, heroin dependence and Medication-Assisted Treatment (MAT)

To better understand the role of law enforcement in facilitating improved access to medical opioids, it is helpful to have a clear understanding of contextual issues relating to heroin dependency, MAT and opioids for pain relief.

### Understanding heroin dependence

A sound understanding of the basics of heroin dependency provides context for better medical management – including for offenders and prisoners who are heroin dependent – and for appreciating the role of MAT.

Drug dependency is a complex condition encompassing a broad range of physical, psychological and social aspects. Clinical guidelines for a definite diagnosis of 'dependence' are detailed in the WHO's International Classification of Diseases (or ICD-10), and require that three or more of the following six characteristic features have been experienced or exhibited:

- a strong desire or sense of compulsion to take the substance;
- difficulties in controlling substance-taking behaviour in terms of its onset, termination or levels of use;
- a physiological withdrawal state when substance use has ceased or been reduced, as evidenced by – the characteristic withdrawal syndrome for the substance, or use of the same (or a closely related) substance with the intention of relieving or avoiding withdrawal symptoms;
- evidence of tolerance, such that increased doses of the substance are required in order to achieve effects originally produced by lower doses;
- progressive neglect of alternative pleasures or interests because of substance use, increased amount of time necessary to obtain or take the substance or to recover from its effects;
- persisting with substance use despite clear evidence of overtly harmful consequences, such as harm to the liver through excessive drinking, depressive mood states consequent to periods of heavy substance use, or drug-related impairment of cognitive functioning efforts should be made to determine that the user was actually, or could be expected to be, aware of the nature and extent of the harm.<sup>18</sup>

The majority of people who use illicit opioids (particularly heroin),<sup>19</sup> or who are prescribed opioids for pain relief,<sup>20 21 22</sup> do not become drug dependent. However, if heroin is used persistently for a significant period of time, some people may develop drug dependency. When people use heroin consistently and over time, adaptation in the body occurs in efforts to stabilize to usual levels. The body stops producing its own opioids (called 'endorphins') in an effort to regain the usual balance. This explains why heroin dependent people report needing to take increasing doses to achieve the same effect (i.e. 'tolerance').

Published data of opioid use in the treatment of pain reveals only a small risk of opioid dependence in patients who had no history of illicit substance use.<sup>23</sup> <sup>24</sup> <sup>25</sup> Tolerance is characterised by a need for markedly increased amounts of the substance to achieve intoxication or the desired effect, and a diminished effect with continued use of the same dose. If a dependent person suddenly stops taking heroin, it takes his or her body a few days to return to normal endorphin production levels. In the meantime, with no heroin or endorphins in the body, the person will experience an uncomfortable period known as 'withdrawal'. The degree of discomfort correlates with the degree of dependency or the size and frequency of a person's usual dose.

While many people may successfully overcome the physical withdrawal symptoms associated with the cessation of heroin and opium use, the complex nature of drug dependency means that they may also experience a range of psychological cravings for some time. This is why 'relapse' can be common for many heroin dependent people. In the context of pain management, unrelieved pain can also precipitate relapse.

### What is Medication-Assisted Treatment (MAT)?

Medication-Assisted Treatment, using long-acting drugs such as methadone or buprenorphine,

is currently the most effective treatment available for heroin dependence<sup>2</sup>. Medicines used for dependency therapy, such as methadone and buprenorphine, have sometimes been labeled Opioid Substitution Therapy or OST. It is important to note that 'substitution' has sometimes been misunderstood and that this has contributed to stigma against, and reluctance to provide, such therapy. In fact, heroin and MAT medications have completely different pharmacological properties. Some scientists prefer to use the term Long-Acting Opioids (L-AOs) or simply 'maintenance agents', highlighting the pharmacological differences between L-AO medications

<sup>&</sup>lt;sup>2</sup> "Drug dependence" is used throughout this paper. In order to understand the difference between "addiction" and "drug dependence," it is necessary to briefly review the history of the evolution of the concept of "drug dependence." During the 1960s, the WHO Expert Committee on Addiction-Producing Drugs made serious attempts to clarify the difference between "addiction" and "habituation," only to abandon this effort and to propose instead the use of the term "drug dependence." In the minds of some experts, this led to the misunderstanding that the meaning of the then new term "dependence" would be the same as "addiction" or "habituation," or both of them combined. This was not the case. As emphasized by that Expert Committee, the term "dependence" carried no connotation of the degree of risk to public health. This was a major difference from the term "addiction," which did carry such a connotation. The same Expert Committee also recommended against efforts to distinguish between "physical dependence" and "psychic dependence," because it felt that all drug effects on the individual are potentially understandable in biological terms. In addition, the Committee noted that "physical dependence" had been confusing to some clinicians because the manifestation of withdrawal syndrome was interpreted as evidence of both "physical dependence" and "drug dependence." From World Health Organization (WHO). *Narcotic and Psychotropic Drugs - Achieving Balance in National Opioids Control Policy:* Guidelines for Assessment. WHO: Geneva. 2000.

and shorter-acting opioids such as heroin.<sup>26</sup> For the purposes of this document, the American terminology of MAT or'Medication-Assisted Treatment has been utilised.

Methadone and buprenorphine are the two most commonly used MAT medications for treating heroin dependence. Buprenorphine is at least three times more expensive than methadone, but can be provided in a drug combination form with naloxone (under the trade name Suboxone) which is thought to reduce drug diversion.<sup>27</sup> Within one or two weeks of beginning MAT, most heroin users experience reduced craving, and over a period of time decrease or stop their use of heroin. MAT offers physical stability in contrast to the often dramatic highs and lows that are associated with dependent illicit heroin use.

There are a number of reasons why people who are dependent on drugs such as heroin may be motivated to seek methadone or buprenorphine treatment, including:

- to reduce, and ultimately eliminate, illicit drug use;
- to re-establish their lives and become more functional members of their community;
- due to an acute crisis in their lives;
- for added stability to assist in undertaking HIV or hepatitis C (HCV) treatment;
- to eliminate financial, physical and social costs involved with heroin use and related lifestyle; and/or
- as a requirement of the criminal justice system in their jurisdiction, by way of early referral into treatment programmes.<sup>28</sup>

MAT helps to introduce stability and removes the stress associated with needing to obtain heroin. The risk of contracting blood-borne viruses such as HIV, hepatitis B/C and other harms associated with injecting are reduced. Overall, the goal of these treatments is to improve the health, social and economic outcomes for individual drug users, their families and the community. In addition, people receiving MAT report less criminal activity, improved family ties, fewer attempts at suicide, less criminal convictions and assisted adherence to HIV medication. In America, where much research on MAT has been conducted, it has been repeatedly demonstrated that 80% of people retained in MAT programmes will reduce of eliminate their involvement in crime.<sup>29</sup>

Of particular significance is the length of time in treatment and the dose. The longer that a person is in treatment, the less likely they are to use heroin and, therefore, the better the outcome. Individuals maintained on adequately high doses that completely block withdrawal symptoms, as determined by a medical practitioner, stay in treatment longer and use less heroin than those individuals on lower doses. MAT is a long term rather than a short term treatment. While people do eventually cease MAT, if they do so before they are ready, there is a high likelihood that they will relapse to illicit drug use. Evidence suggests that the minimum period for effective methadone treatment is 12 months, but for some people treatment may be life-long. Therefore, remaining on methadone is not seen as an indication of failed treatment.<sup>30</sup>

The use of MAT is supported by the UN system as an essential element in the management of heroin dependence and the prevention of HIV infection among injecting drug users, and MAT medications are listed by the WHO as 'Essential Medicines'.<sup>31</sup> The INCB has pointed out that MAT "... does not constitute any breach of treaty provisions, whatever substance may be used for such treatment in line with established national sound medical practice."<sup>32</sup>

### **Opioids for pain relief**

The Single Convention on Narcotic Drugs of 1961 highlights that the medical use of narcotic drugs continues to be indispensable for the relief of pain and suffering, and that adequate provision must be made to ensure the availability of narcotic drugs for such purposes.<sup>33</sup> Further, the Convention on Psychotropic Substances of 1971 encourages member states, where necessary, to educate regulators and health-care professionals, including through targeted awareness-raising campaigns, to recognize that the medical use of narcotic drugs continues to be indispensable for the relief of pain and suffering and that adequate provision must be made to ensure the availability of narcotic drugs for such purposes.<sup>34</sup> There is an international medical and scientific consensus that opioids are safe and effective for the relief of pain and treatment of dependence when prescribed by knowledgeable practitioners. Although experienced and reported differently by individuals, pain is a real phenomenon in the nervous system and its treatment should be taken seriously as it can seriously undermine quality of life. Not all pain needs opioids, but in some cases opioids are indeed the primary treatment.

It has been estimated that 6 million people suffer from severe untreated pain resulting from cancer and HIV, with millions more suffering in a similar fashion due to chronic illness, accidents and injuries, and the aftermath of surgery.<sup>35</sup> Palliative care for these patients is needed, but is generally in short supply everywhere. In many jurisdictions, the opioid medicines required to effectively manage and treat pain are unavailable, or remain largely inaccessible to those in need. Many of the laws and regulations presenting legal impediments were adopted before substance dependence and the beneficial medical uses of opioids were understood. Weak health care systems, insufficient training, stigmatisation of opioid use, and cost and distribution problems have combined with legal impediments to contribute to the situation of poor medical opioid access.<sup>36 37 38 39</sup> The impact of these factors upon people who suffer from severe untreated pain is particularly pronounced in developing and transitional countries.<sup>40 41</sup>

Many national drug policies reflect disproportionate concerns that the use of opioid medications in pain management and palliative care will be diverted to illicit markets and result in patients becoming opioid dependent. While efforts to address those who illegally divert controlled medicines are a legitimate target of law enforcement, some policies have had unintended negative effects on patient care and medical practice. Physicians are also deterred by the fear of stigmatisation and professional discipline, criminal sanction or even imprisonment for mistakes in opioid dispensing or record keeping.<sup>42 43 44 45</sup>

Systems of control and outdated understandings of drug dependency have unfortunately interfered with safe patient access to medical opioids and defeat the purpose of the Conventions. Laws and law enforcement practices have hampered the goal of medical availability. A streamlined approach, based on the international drug conventions, is required in order to adequately address the complexity of medical opioid access.<sup>46 47</sup> Strategic partnerships which allow the health and welfare sectors to interact with law enforcement are also required.<sup>48</sup> Experiences from a number of jurisdictions that have recognised the need for improved access to medical opioids, and have succeeded in achieving balance between law enforcement and good medicine, have been synthesised for reference throughout Sections 2 to 5 of this document.

In the following sections, different law enforcement approaches which can improve legitimate access to medical opioids and MAT (following the supply chain from cultivation and manufacture through to end user access) are explored and guidance points provided.

# 2 Ensuring the safe manufacture, import, transport, and storage of medical opioids

This section summarises UN Drug Convention requirements relating to the manufacture, import, storage, and transport of medical opioids, and then examines the experience of selected countries, highlighting both negative and positive experiences. This section is of particular significance for drug regulators, with implications for the law enforcement role in a more balanced regulation of the medical opioid (i.e. medicines for both pain relief and for MAT) supply chain.

### A. Summary of Drug Convention requirements

"The Single Convention is the result of the recognition by the United Nations of the fact that the adequate provision of narcotic drugs for medical purposes is indispensable for the welfare of mankind, as well as of the fact that drug addiction is a worldwide social and economic threat. Therefore, the Single Convention aims to restrict the use of narcotic drugs to medical and scientific purposes and to prevent their diversion and abuse, while at the same time ensuring their availability for legitimate purposes. It includes control measures over the cultivation of plants that serve as sources of raw material of narcotic drugs, provisions regarding the obligations of national authorities in the application of control measures over the production, manufacture, trade, and distribution of narcotic drugs, as well as provisions for the medical treatment and rehabilitation of addicts."<sup>49</sup>

The Convention system is intended to ensure medical opioid availability and prevent diversion. Parties are required to nominate a Competent Authority which is authorized to estimate medical requirements, manage import and export licenses, report required statistics, and supervise adequate controls over distribution. In some jurisdictions, the Competent Authority is supervised by agencies that are part of law enforcement.

As most countries do not cultivate opium poppy plants, the importation of opioids is the first – and most stringently controlled – step in the medical opioid supply chain. Operational requirements of control for import and export involve:

- providing details of the competent authority to the INCB;
- licenses for import and separate authorisation for each import;
- importing only from INCB authorised countries;
- providing detailed import certificates to the competent authority of the export country to secure export authorisation;
- having the export certificate referencing the import certificate together with the export and a copy is provided to the importing country;
- specifying the amount actually exported on the export authorization;
- recording completion and amount of each import and reporting this to the exporting country; and,
- quarterly documentation of all imports and exports for the INCB.<sup>50 51</sup>

In practice, the import regulation requirements set forth in the Single Convention have been effective in preventing diversion from international trade.<sup>52</sup> However, the regulatory

requirements do pose challenges at national level where the process can cause supply delays. Countries committed to ensuring adequate availability of medical opioids and MAT, including resource constrained countries, can avoid such delays by establishing an effective streamlined system and running checks to ensure that no blockages occur.

According to the Single Convention on Narcotic Drugs, drug manufacture and production are defined as follows: Drug 'manufacture' means all processes, other than production, by which opioids may be obtained and includes refining as well as the transformation of poppy products into other opioid medications. Drug 'production' refers to the separation of opium from poppy plants.<sup>53</sup>

**Cultivation:** The Single Convention on Narcotic Drugs specifies that member countries applying to cultivate poppy plants for opium production must establish one or more government opium agencies to oversee set requirements relating to designated growing areas, cultivator licensing harvesting and trade (*Article 23*). The Convention also requires that countries notify the INCB if planning to initiate or increase poppy cultivation for the purposes of opium production. Planning for increases in opium production must account for prevailing world need for opium, so that the production of opium does not result in over-production. If production is likely to result in illicit trafficking, production should not occur (*Article 24*).<sup>54</sup>

**Manufacture:** Member countries must license manufacturers with periodical permits and ensure that excess accumulation does not occur (*Article 29*). The Convention also requires countries to refrain from producing opium or manufacturing opioids if doing so may result in illicit traffic.<sup>55</sup>

**Storage and transport:** Monitoring and control of domestic storage and transport of opioid poppy products is not detailed in the Convention, so designated government opium agencies are required to devise their own systems. However, at every stage of the supply chain, *Article 33* on possession of drugs applies: "The Parties shall not permit the possession of drugs except under legal authority."<sup>56</sup>

Provisions of the Single Convention regulating dispensing and administration of opioids involve relatively straightforward licensing and medical prescription safeguards (discussed in Section 3 below).

Beyond the specifications summarised above, specific domestic operational policies and practices connected with licit opioid production are the responsibility of member countries. This means that countries have considerable flexibility (as well as obligation) to apply reforms around unnecessarily restrictive drug control laws, regulations and practices to correct the medical opioid control and availability balance.

### B. Control to ensure availability: experience from selected countries

The overarching thrust of the Single Convention on Narcotic Drugs instructs countries to ensure access for medical purposes, while at the same time ensuring diversion control of licit opioids. Most jurisdictions have tended to over-emphasise control at the expense of access, thus impeding the Convention's mandate and causing unnecessary pain and suffering to those in

need. The Convention is clear that while the prevention of diversion is important, it should work together with measures to ensure adequate availability of opioids for medical purposes. The Convention's diversion control specifications are set out so that member countries have the flexibility to implement them in ways which accord with the Convention mandate, as well as with domestic needs and resources. Here we explore the experience of three countries – Romania, Australia and the United Kingdom – drawing out lessons on flexibility to increase the availability of opioids for legitimate purposes without overly compromising diversion control. These lessons can be extrapolated to different country contexts.

The strict regulation that is characteristic of over-emphasis on diversion control often leads to regimes in which the rules for manufacturing or importing controlled medications, and transporting them to pharmacies, make routine end user access difficult or impossible. When excessively burdensome regulations are applied, they can cause blockages in the supply chain.

#### Example from Romania

Romania has made great strides forward in narcotics law reform and the implementation of a new regulatory system capable of providing greater capacity at the prescription end of the medical opioid chain.<sup>57 58 59</sup>

While acknowledging progress, some regulatory storage impediments remain. The Single Convention specifies only that storage must be managed under legal authority, and many countries do not impose licensing or further requirement beyond lock and key security. However, in Romania, licenses are required to store opioids. To apply for a storage license, the applicant must provide:

- a completed application;
- a registration number;
- a warehouse license;
- a curriculum vitae of the pharmacists who may interact with the substances;
- the criminal record of the pharmacists involved with the substances; and
- a statement of diversion prevention measures to be undertaken.<sup>60</sup>

### Example from Australia

The Tasmanian poppy industry has been active for more than 30 years and supplies about half of the world's medicinal opioid market. As well as cultivation, initial processing is undertaken in Tasmania and the neighbouring state of Victoria.<sup>61</sup>

The Australian federal government and the Tasmanian state government share responsibility for control of the poppy industry, which is undertaken effectively via cooperation between regulatory and law enforcement authorities. The Single Convention on Narcotic Drugs requires control and supervision of poppy growing and production. Fulfilling this regulatory role, the Poppy Advisory and Control Board advises upon matters relating to the cultivation, production and transport of Tasmanian poppy, liaises with Australian government departments and oversees security matters for Tasmanian poppy crops.<sup>62</sup>

Security issues are coordinated between the Poppy Advisory and Control Board, licensed growers and manufacturers, and law enforcement. Strict controls are maintained over all aspects of poppy cultivation and processing, with Poppy Advisory and Control Board field officers regularly patrolling crops and liaising with the Tasmanian Police Poppy Task Force,

which operates each poppy season.<sup>63</sup> During the poppy growing and harvesting season, crops are regularly monitored by Poppy Advisory and Control Board field officers, with any illegal activity investigated by Tasmania Police Poppy Task Force officers.

Trespassing on poppy fields and the unauthorized possession of poppies and poppy products is illegal, with substantial penalties for offenders. According to Hobart's *The Mercury* newspaper, there were 17 thefts of poppy heads during 2008-09. These figures are seen as relatively low and not a cause for concern.<sup>64</sup> All persons connected with the industry undergo security checks (scanning for criminal history and, in particular, drug related offences) with assistance from police intelligence. Access to poppy fields is restricted to authorised individuals, such as license holders and their employees, company field officers, and employees contracted during sowing and harvesting times. Storage of poppy products is controlled via secure facilities fitted with cameras.<sup>65</sup> Transport of poppy products from growers to processing companies is regulated as part of existing licensing arrangements.

The Australian poppy industry provides a model demonstrating a partnership approach as part of the role of law enforcement in ensuring safe manufacture, storage and transport of medical opioids. Security checks for manufacture through to transport are provided via collaboration between state and federal governments, an especially commissioned Control Board, licensed growers and manufacturers, and law enforcement.

### Example from the United Kingdom

The United Kingdom (UK) provides an interesting variation of regulation of the transport of medical opioids. The UK government does not provide specific regulations for licit opioid transportation, but does provide guidelines and administer individualised inspections on an ad hoc basis. The guidelines were created in consultation with the pharmaceutical and shipping industries, and are designed to support transporters to formulate their own control plans by addressing record keeping, checks and reporting.<sup>66</sup>

The UK approach demonstrates how government flexibility in implementing regulatory systems can expedite the supply process. The UK government has acknowledged that some level of diversion is inevitable, but that such is not cause for extreme concern and disproportionate measures. The guidelines are generally framed as recommendations, rather than strict and cumbersome requirements. For example, it is advised that whenever diversion of medical opioids in transit occurs, the consignor and the consignee should immediately review their procedures to prevent recurrence. Mishandling has been a problem at transit points, so it is suggested that shipping agents use the most direct route in order to minimise the opportunities for mis-routing or diversion. In the past, when a consignment has gone missing, it has been the practice of some organisations to assume that it has been misdirected and to wait before reporting. Currently, however, if the consignment is not received at the expected time, it is advised that the supplier should be advised without delay, and the supplier should then promptly report the matter to the carrier, the police and the Home Office.<sup>67</sup> The UK medical opioid transportation system relies mainly on self-regulation by the transport industry and appears to provide a practical and functional framework for facilitating the safe and efficient transport of medical opioids.<sup>68</sup>

### Guidance points

Drug regulations should be analysed to ensure the identification and removal of potential bottleneck points and onerous licensing requirements. Healthcare providers should participate in reviews of drug control regulations to assess whether they unnecessarily impede accessibility of pain medications. If regulations are found to impede access, they should be amended.<sup>69</sup> Regulators should ensure an effective supply chain for opioid medicines with regulations or guidelines that address key areas of potential diversion but do not unduly complicate the supply chain. In this way, drug control laws and regulations must acknowledge the essential nature of opioid medications for the relief of pain and suffering so as to ensure access for end users. Law enforcement efforts can then become more targeted, focusing largely on key diversion sources, and thereby facilitating a fluid and functional supply chain.

Governments have much flexibility in how to manage the medical opioid and MAT supply chain to satisfy UN Drug Convention requirements. Cultivation and manufacture require application, central oversight and licensing, while storage and transport must be under 'legal authority' of some description. The UK experience relating to production, storage and transport provides an interesting example of streamlining access to medical opioids by simplifying regulations.

Balanced, proportionate diversion methods must take into account that medical opioids and MAT are essential medications, and that while potential diversion should be identified and averted, the potential illicit activities of a few cannot be allowed to interrupt or delay sorely needed medical provisions for many.

# 3. Diversion control: not hindering access by prescribers and dispensers, and by end-users, to medical opioids

Responsible control is required to limit the diversion of medical opioids. However, balance must be achieved so that control does not restrict the prescription and subsequent consumption of necessary medical opioids. According to the Conventions, national policy should specify that opioids are indispensable, that governments must ensure adequate provision as well as diversion control, and must establish a competent authority to oversee implementation of Convention requirements. It is important to note that this authority is often connected with law enforcement. Therefore, the onus for striving towards an effective balance between medical opioid and MAT availability and diversion control, often rests with law enforcement authorities in particular. This subject is explored in greater detail in this section on medical opioid access by prescribers, dispensers and end users.

The Single Convention on Narcotic Drugs lays out three minimum criteria that countries must observe in developing national regulations regarding the handling of opioids:

- individuals must be authorized to prescribe or dispense opioids by their professional license to practice, or be specially licensed to do so;
- movement of opioids may occur only between institutions or individuals so authorized under national law;
- a medical prescription is required before opioids may be dispensed to a patient.<sup>70</sup>

The INCB has recommended that health professionals should be able to provide opiates without unnecessary fear of sanctions for unintended violations – including legal action for technical violations of the law – that may tend to inhibit prescribing or dispensing of opiates.<sup>71</sup> WHO guidelines stipulate that the policing of opioid prescription should be conducted with well informed sensitivity for the diversity and importance of discretion in physician decision making, ensuring that appropriate access to medical opioids is an explicit goal of law enforcement practice, in balance with control of illegal use.<sup>72</sup>

In practice, many countries have regulations relating to medical opioids (including MAT) that involve unnecessarily complex procedures which ultimately limit access to those who need them. For example, a recent review of medical opioid availability for cancer-related pain management in Europe found that regulatory restrictions to limit the accessibility of opioids were common, particularly in eastern European countries.<sup>73</sup> These control measures mean that pharmacies and health providers do not stock opioids, that doctors do not prescribe opioids because of the effort required or fear of liability, and that opioid prescription processes are so daunting that many of those in legitimate need do not have adequate access.<sup>74</sup>

Many of the regulations regarding opioid prescribing and dispensing contradict WHO and INCB recommendations. Examples include requirements for patients to have a special permit, arbitrary dose limits, limitations on settings where opioids can be prescribed, restrictions on prescribing privileges to limited physician specialties, unwarranted restrictions on the number of days' supply that can be prescribed at one time, and excessive limits on types of opioid dispensing sites.<sup>75</sup> The impact of such over-vigilant restrictions on patient care can be extreme.

It is evident that "the science and best practices of opioids have progressed more rapidly than the legal structures governing them, leaving many antiquated and overly restrictive legal policies."  $^{76}$ 

Fear of diversion drives many jurisdictions' policies regarding medical opioids, and "a default position of limiting or precluding supply of prescription opioids for medical conditions appears to be the norm."<sup>77</sup> This can result in restricted access to essential medications for many people in need. Furthermore, even in countries with an exclusive focus on diversion control, the measures appear to be not only unsuccessful in completely avoiding diversion, but also to contribute to growing epidemics of opioid injecting and HIV transmission, while choking off legitimate patient access. Diversion control measures are often instigated without an evidence base detailing key diversion sources.<sup>78</sup> "As long as fear of diversion exists, and no examination of the situation is made, it is likely that efforts to control diversion will be misdirected and lead to overly restrictive control of supply."<sup>79</sup>

Once key diversion sources are identified, a more proportionate and efficient response can be implemented. Of note, the INCB has also stated that diversion of opioid medications at the international level, from the licit trade into illicit channels, remains relatively rare and in small quantities compared to the large trade flow.<sup>80</sup>

Examples of disproportionately restrictive regulation can be found in most jurisdictions where the implementation of unnecessarily complex licensing processes for healthcare institutions and health workers, and burdensome prescription procedures and prescription limitations, can impede access to adequate pain management and MAT. Fears of legal sanctions also contribute to the under-availability of essential medical opioids, including MAT.

### A. Diversion control: the unintended impact

### Fear of liability

This unintended impact is related to communications problems on the part of regulators, as well as to how law enforcement conducts its investigations. In most jurisdictions, diversion control processes have tended to put pressure on the health sector by threatening prescribers and dispensers with potential legal sanctions,<sup>81</sup> and this has resulted in the under-prescribing of essential opioids.

In the United States, for example, doctors fear unjustified prosecution for prescribing opioids for pain and tend to react with excessive restraint in issuing opioid prescriptions. A 2008 study on prescribing offences in the United States found that, although convictions are relatively rare, mixed messages from regulators have resulted in fear among prescribers, undermining the treatment of pain. The study concluded that "Effective solutions to the conflicting public health crises of under-treated pain and prescription drug abuse will have to address the discordant perceptions between physicians and law enforcement."<sup>82</sup>

While countries are obliged to prevent inappropriate prescription of medical opioids, it is also important that regulators and law enforcement take a balanced approach in public messages to physicians and how they handle routine investigations of medical practice.<sup>83</sup>

### **Prescription monitoring**

While the Single Convention (*Article 30 – 2(b)(ii)*) states that countries may consider requiring official prescription forms in 'counterfoil books' for opioid prescriptions, in several countries that have introduced the 'triple prescriptions' system as part of their approach to controlling the diversion of medical opioids, this has restricted prescribing to patients requiring pain management medications.<sup>84</sup> In most European countries, for example, difficulty in accessing the required prescription is common, and in some countries, physicians need to purchase the prescription forms.<sup>85</sup>

The United States (US) Drug Enforcement Administration (DEA) has declared, without supporting data, that prescribing and dispensing accounted for the majority of prescription opioids diverted to the black market in the United States,<sup>86 87</sup> while other US law enforcement and regulatory agents believe that the major diversion mechanism is 'doctor shopping' (where an individual 'shops' from doctor to doctor, acquiring multiple prescriptions for prescription medications) and pharmacy theft or forgery.<sup>88</sup> Prescription monitoring programmes (PMPs) have been introduced in some American states to prevent and detect the diversion of controlled substances. PMPs are not intended to interfere with appropriate medical practice. However, clinicians' concerns about increased regulatory scrutiny could lead to a reduction in opioid use and, therefore, less than optimal pain control. Further, there has been little research to examine the impact of PMPs on physician prescribing, pain management or drug diversion.<sup>89</sup>

The relative significance of different diversion sources is not well understood. There is much concern about possible diversion mechanisms without indicating respective importance. For example, although sources of diversion of prescription opioids in the United States are not well known, the primary regulatory and law enforcement responses tend to be directed at patients, prescribers and manufacturers. However, some researchers have discovered extensive criminal diversion activities – including significant quantities of opioids stolen from distributors and pharmacies before the drugs are prescribed or dispensed – suggesting that law enforcement diversion control efforts should be appropriately focused on these sources in the US, <sup>90, 91, 92</sup> and that similar research should guide diversion control efforts in other regions.

**Other unintended consequences of diversion control** can include stimulation of the illicit drug market and increased drug related harm. The INCB notes that some countries have drug control laws and regulations with provisions that go beyond convention requirements, without necessarily preventing diversion. In particular, overly stringent prescription requirements may lead to a situation where certain controlled drugs are more readily available on the unregulated market.<sup>93</sup> Other negative impacts can include:

- increased crime as patients and/or drug dependent individuals turn to the illicit market to obtain pharmaceutical preparations, which are more expensive on the black market due to their scarcity;
- substitution of medical opioids with other drugs (i.e. alcohol, illicit drugs, or less effective over-the-counter analgesic medications that can result in liver or gastro-intestinal toxicity) leading to other, potentially more severe, health issues;

 the creation of conditions favorable for increased heroin trade and/or diversion from the medical opioid supply.<sup>94, 95</sup>

It is therefore recommended that diversion sources be scrutinized through a research process which identifies and prioritises key diversion sources for medical opioids. From this information, balanced and proportionate control measures can be introduced with the aim of both limiting diversion and ensuring access for legitimate purposes.

### The law enforcement impact on access to MAT

People participating in MAT programmes are frequently required to undergo mandatory urine testing, punitive dose reduction and discontinuation of treatment (involving withdrawal). Raids, arrests and harassment in or near MAT clinics have been reported in many countries including Malaysia, Kazakhstan, China, and Thailand.<sup>96</sup> During 2007 in Odessa, Ukraine, patients in a MAT support group experienced ongoing harassment from law enforcement. In Kazakhstan, opposition from the Ministry of Interior has delayed the implementation of MAT programmes for years. In Kyrgyzstan, police threaten methadone patients with arrest or demand bribes.<sup>97</sup>

People receiving MAT in community clinics often have their names and personal details added to government registries of drug users. For example, in Indonesia, it is reported that police are not well informed about the legality of MAT (in this case, methadone) and continue to make arrests in the vicinity of MAT clinics. As a result, some MAT clients report that they have become targets for the police.<sup>98</sup> This experience highlights the need for law enforcement to adopt and implement policies, practices and ongoing education that support the medical treatment of drug users such that MAT can be more accessible to those requiring it.

### MAT regulations

MAT entry protocol changes are crucial if treatment is to be accessible to those in need. For example, until recently, China required that people using drugs undergo up to one year of detention in compulsory detoxification facilities or forced labour camps prior to entry into a methadone programme.<sup>99</sup> Law enforcement often held the right to grant or deny admission into MAT programmes. Although these requirements were recently relaxed, some MAT clinics are yet to fully comply. For entry into MAT programmes, current Chinese guidelines stipulate four conditions, including that the person seeking MAT must:

- 1. pass through drug detoxification programmes multiple times;
- 2. be more than 20 years of age;
- 3. be a resident of the county, city or district where the treatment is provided (or have a temporary residence permit); and
- 4. exhibit 'civilised' behavior.

The age limit is waived if the person seeking MAT is HIV positive.<sup>100</sup> <sup>101</sup> Only those with residence permits issued by police are eligible – and these are routinely denied to migrants or others.<sup>102</sup>

In Georgia, people seeking entry into MAT programmes must document that they have previously participated in a 'drug free' treatment programme. 'Drug free' treatment remains both expensive and scarce, and those individuals who undertake such treatment are added to government drug user registries, with the potential for harassment and discrimination.<sup>103</sup>

MAT regulations must be sufficiently straightforward and confidential, so as to make MAT accessible to those who stand to benefit from the therapy.

### **B.** Important considerations towards developing diversion control regulations, policies and practices

Law enforcement is generally bound by drug control regulations, which can be seen as restrictive by prescribers and dispensers. WHO guidelines stipulate that the policing of opioid prescription should be conducted with well informed sensitivity for the diversity and importance of discretion in physician decision making, whilst ensuring that appropriate access to medical opioids is an explicit goal of law enforcement practice, in balance with the control of illegal use.<sup>104</sup>

*Article 38* of the Single Convention, 'Measures against the Abuse of Drugs', provides some direction with regard to addressing illicit drug use and treatment of dependency. Paragraph 1 directs member states to give special attention to the treatment, education, rehabilitation, and social reintegration of people involved in illicit drug use, and paragraphs 2 and 3 specify that there should be adequate training and education of personnel in order to achieve the objectives of paragraph.<sup>105</sup>

However, even where facilitating policy and regulatory environment exists, cases of obstructive police activity are not unknown. Policing needs to be consistent with regulatory reforms in order to control diversion while supporting the expansion of medical opioid access.

An Australian study found that, from the perspective of law enforcement personnel, the policing of diverted medical opioid markets posed particular challenges. Key issues related to such policing include:

- the difficulties in distinguishing between illicitly and licitly held prescription pharmaceuticals (i.e. pharmaceutical identification);
- becoming aware of relevant scheduling and legislative considerations;
- developing an understanding of psychopharmacology of benzodiazepines and prescribed opioids, interactions with illicit drugs, and implications for behaviour;
- the apparent weaker relationship between prescription pharmaceutical use and crime than for illicit drugs; and
- similar policing responses were required regardless of whether intoxication is due to use of licit or illicit drugs.<sup>106</sup>

A potential range of responses to the challenges associated with policing diverted medical opioid markets includes:

- the creation of alternatives to arrest and criminal charges, possibly through liaison with diversion programmes and service providers;
- decreasing the cost of drug treatments;
- a more holistic approach to the prescribing of drugs;
- the close monitoring of people who inject drugs who are prescribed central nervous system depressants;
- the development of alternative forms of buprenorphine that cannot be diverted;

- keeping police and doctors informed about prescribed drugs that are likely to be diverted;
- the education of doctors and pharmacists about the diversion of such drugs;
- encouraging the sharing of information between different bodies that produce data;
- peer education programmes built around demonstrating the harms associated with the intravenous administration of tablets;
- the distribution of pill and biological filters (used to 'filter out' non-soluble contents in pharmaceutical preparations) through needle and syringe programmes (NSP) in order to reduce health harms; and
- the establishment and maintenance of close relationships between the health and law enforcement sectors.<sup>107</sup> <sup>108</sup>

The development and relative priority of responses to be implemented should be informed through research to identify key diversion sources, and balanced against the relative significance of the source, and possible unintended consequences.

Finally, a review of terminology used by drug regulators is recommended. Several countries use stigmatizing terms for opioid analgesics (such as 'drugs of addiction', 'dangerous drugs' or 'poisons') in the regulations controlling their prescription and dispensation.

### Guidance points

It should be recalled that, at the international level, diversion of opioid medications from the licit trade into illicit channels remains relatively rare and in small quantities compared to the necessarily large trade flow.<sup>109</sup>

Medical opioid diversion control programmes usually have three goals:

- 1. to limit access to those individuals with a legitimate need for the drug;
- 2. to identify and track instances where control over such legitimate access is compromised; and
- 3. to minimise the effect of controls upon legitimate medical practice.

These three goals should be balanced to inform contextually specific strategies.<sup>110</sup>

Research should guide diversion control efforts to be proportionate and relevant to the significance and characteristics of the source, thus promoting improved flow in end user access to opioid medications, including for MAT. Once key diversion sources are identified, a more proportionate and efficient response can be implemented. Measures which do not restrict, delay or interrupt legitimate access to opioid medications can then be implemented and evaluated.

With regard to reducing misplaced stigma attached to the prescription and use of medical opioids, it is important that regulators and law enforcement learn and understand the need for medical opioids, including MAT, and take a more balanced approach in public messages to physicians and how they handle routine investigations of medical practice.<sup>111</sup>

The requirement for special prescription forms is not necessarily excessive, provided that forms are made readily available to prescribers.

PMPs should not be administered by law enforcement agencies, but rather should be seen as public health intervention tools. Guidelines in operating prescription monitoring include:

- protecting patient confidentiality;
- assuring individual healthcare professionals access to monitoring data about their individual patients, so that they can evaluate those patients' use of controlled substances;
- allowing law enforcement agencies access to the data, but only when probable cause justifies such access in the course of investigating possible abuse or diversion;
- developing educational programmes to minimise concerns about regulatory scrutiny when prescribing or dispensing controlled substances as part of legitimate medical practice.<sup>112</sup>

Drug regulators should avoid pejorative terms for opioid analgesics in the regulations controlling their prescription and dispensation.

Planning for medical opioid diversion prevention must be carefully considered and balanced against potential unintended consequences. Laws and regulations must be tailored to the needs and responsibilities of all stakeholders, so as to ensure access without excessive restrictions or the deprivation of necessary treatment. It is possible that significant increases in the availability of opioid pain medication will be accompanied by increased diversion of prescription opioids, but that these problems can be managed by a combination of provider training, patient education and regulatory oversight. The social benefits of better access to essential opioid medications clearly outweigh the costs.<sup>113</sup>

### MAT specific guidance

People accessing MAT programmes are among the most marginalised in society and, particularly when not in treatment, vulnerable to HIV transmission. MAT has proved to be an essential component in an effective public health response and should therefore be unobstructed by any extraneous police activity. There is a clear need for law enforcement to adopt and implement policies, practices and attitudes that support the medical treatment of drug users.

- MAT regulations must be sufficiently straightforward and confidential, so as to make MAT accessible to those who stand to benefit from the therapy;
- Police should not routinely make arrests or question clients in or near MAT clinics or dispensing areas. This does not mean that police should not respond to reported criminal activity wherever it may occur – but does mean that police should not use MAT clinics to fill arrest quotas, or for over-patrolling;
- It is not a requirement of the Single Convention on Narcotic Drugs that police keep records on MAT clients. Police must not routinely keep records on MAT clients or access clinic records.

# 4. Law enforcement assisting access by end-users to medical opioids

In this section some examples of law enforcement working to enable improved medical opioid access are provided, in both the areas of pain relief and MAT. In the pain relief context, drug regulators frequently play a vital role in removing unnecessary obstacles to patient access to medical opioids. The role of law enforcement in the Chinese regulatory reform experience is examined. In the context of improving access to MAT, examples are provided of law enforcement forging linkages with community drug treatment providers, and diverting people who use drugs to drug treatment.

It should be noted that diversion, in this section, refers to the range of models involving the process of diverting a drug-related non-violent offender from the criminal justice system to non-custodial education and/or treatment.

### A. Working partnerships between law enforcement and health agencies (including hospitals / clinics providing pain relief and MAT providers)

Some jurisdictions have experienced successful medical opioid access reform with the law enforcement sector playing a central role.<sup>114</sup>

### Medical opioids for pain relief – an example from China

Law enforcement agencies - working with public health authorities and with support from international community resources – have played a pivotal role in the process of policy reform to reduce legal barriers to pain relief in China. The Ministry of Public Security authorised a process of assessment, policy reform and implementation that reduced the legal barriers to pain relief.

Information was provided and applied workshops were held to address traditional negative attitudes towards opioid medications among government officials and delegates of legislature. Concepts such as drug tolerance and drug dependence were clarified, and exaggerated fears relating to opioid dependence have decreased. Reform changes encouraging improved opioid medication supply included:

- the Chinese Food and Drug Administration (FDA) became the central agency for production and distribution safety;
- simplified procedures for the production, pharmaceutical management and selling of medical opioids;
- relaxation of restrictions on production, storage, and shipment; and
- restructuring of prescription licensing procedures.<sup>115</sup>

Substantial increases in consumption of medical morphine demonstrate the positive impact of the reform process in China. Training on the use of opioid medications in cancer care is now provided through institutional infrastructure as a requirement for prescribing physicians, and a recent survey has shown improvement in basic knowledge for those physicians who have receive training in cancer pain management.<sup>116</sup> However, fear of opioid dependency continues to dissuade physicians from prescribing morphine,<sup>117</sup> and some barriers in China are yet to be

addressed, such as time limits on opioid prescriptions, improving opioid medication access for rural patients, and dispelling stereotypes about palliative care in cancer treatment.<sup>118 119 120</sup>

Input from the Chinese Ministry of Public Security helped to establish balance between law enforcement concerns and more streamlined access to opioid medicines for those in need. This example demonstrates a role for drug regulators and law enforcement agencies in not hindering, but instead facilitating, improved access to medical opioids.

### Partnerships between law enforcement and drug treatment (including MAT) providers

In addition to not hindering access (*as discussed in Section 3*) law enforcement plays a central role in optimizing access to, and benefits from, MAT.

A systematic review of drug law enforcement evaluations examined a range of law enforcement approaches, including:

- international / national interventions (i.e. interdiction and drug seizure);
- reactive / directed interventions (including crackdowns and raids);
- partnership interventions (i.e. problem-oriented policing and community policing);
- individualised interventions (arrest referral); and
- interventions that used a combination of reactive / directed and partnership strategies.

Results indicated that "... proactive interventions involving partnerships between the police and third parties and/or community entities appear to be more effective at reducing both drug and non-drug problems in drug problem places than are reactive / directed approaches."<sup>121</sup>

Law enforcement can enhance the effectiveness of MAT by indirectly or directly facilitating access.<sup>122</sup> MAT is supported as an essential public health service by law enforcement through working partnerships with MAT (and other drug treatment) programmes in some jurisdictions. Part of the basis for such law enforcement collaboration with the health and welfare sectors is the evidence demonstrating that criminal behaviour, particularly for property and drug offences, progressively reduces the longer that an individual remains on MAT, and that offending is often high among those individuals who leave such treatment prematurely.<sup>123</sup>

One example of successful collaboration between law enforcement and the health and welfare sectors is found in Kolkata, India. Drug law enforcement in India is bound by the Narcotic Drug and Psychotropic Substance Act of 1985. However, the police in Kolkata have adapted drug control programmes to reflect current needs. Further, law enforcement in Kolkata are well informed regarding harm reduction due to close ties and working relationships with several NGOs operating in drug use fields.<sup>124</sup> The Society for Community Intervention and Research (SCIR), with the assistance of Sharan Society for Service to Urban Poverty, established two drop-in centres for people who inject drugs, during the late 1990s. In 1999, a three year trial MAT programme was implemented via these drop-in centres. The smaller of the two drop-in centres is located directly beside a police station. Collaboration with the Kolkata Police (North Division), including patronage by the Kolkata Deputy Commissioner of Police, has ensured that law enforcement and health and welfare efforts are complimentary, to the point that the police

stationed there frequently make tea for, and occasionally supply food to, SCIR staff and those people using drugs accessing the drop-in centre.<sup>125</sup>

Law enforcement and health workers should collaborate in order to ensure the effective delivery of MAT programmes. To establish partnerships, law enforcement and the staff of MAT programmes need to create routine communication mechanisms to ensure regular dialogue on emerging challenges and issues. Information that may be useful for law enforcement and the staff of MAT programmes to share can include:

- review of the public health objectives and proven benefits of MAT;
- ground rules regarding information sharing, which comply with the codes of ethics and objectives of both services (i.e. respecting confidentiality);
- specific information about the roles of the staff involved in the MAT programme;
- the location of MAT clinics and dosing sites;
- the hours of operation of MAT services;
- the identification of main contact persons within law enforcement and the MAT programmes.<sup>126</sup> <sup>127</sup> <sup>128</sup>

### Police concerns about MAT

Police have sometimes raised concerns in relation to MAT programmes. The following examples are adapted from New South Wales (NSW) Police Guidelines:

### • MAT programmes attract illicit drug users to the area

It is important to remember that persons attending MAT programmes are illicit drug users who have opted to undertake legal treatment in an effort to solve problems associated with their illicit drug use. Generally, individuals attending clinics will live or work within the area – this is to facilitate access to, and compliance with, treatment.

At times, clients of MAT programmes may be involved in illegal activities. Balancing public order and public health concerns is not always easy, but MAT programmes are a public health strategy designed to reduce the aggregate harm of illicit drug use to the wider community.

### • MAT programmes attract drug users, causing increased levels of crime and public amenity problems

MAT programmes are generally established in response to treatment needs in a given location. Research shows that the presence of MAT clinics does not necessarily increase localised crime. If, however, police are concerned about illegal activity – particularly drug dealing – in the vicinity of MAT sites, they should seek to resolve this through liaison with the manager of the MAT programme.

### • Drug dealers target MAT programmes

MAT clinics are not off limits to police. If the supply of drugs and/or other criminal behaviour is occurring in the vicinity of a MAT site, police should take appropriate action. Where possible, police should consider liaising with the management of the MAT programme beforehand.<sup>129</sup>

### • How can police deal with public criticism of MAT programmes?

Police often feel caught in the middle of the wide range of community opinions regarding MAT programmes. It can be helpful if police officers are familiar with the evidence supporting MAT (*as outlined in Section 1D of this guidance tool*).

## B. Pre-arrest referral of people who use drugs illicitly to health and welfare groups

Working partnerships between law enforcement and drug treatment (including MAT providers) are becoming an increasingly legitimate category of law enforcement intervention through caution and arrest referral (i.e. diversion) schemes. These approaches are supported by the UN Drug Conventions and the United Nations system. See, for example, the 2007 report of the INCB on proportionality in dealing with drug-related offences.<sup>130</sup>

In India, the Calcutta Samaritans deliver regular sensitisation workshops to law enforcement on the subject of drug use and drug treatment. The result of this ongoing collaboration includes an informal arrangement which involves first time drug offenders being referred for drug treatment, rather than being directed into the criminal justice system.<sup>131</sup> Law enforcement is actively involved with the NGOs, with at least one officer in each police station sufficiently motivated and aware of the present drug use scenario (and associated issues). Experience has helped to reinforce positive perceptions of harm reduction approaches, including MAT, as law enforcement and the broader community have observed the encouraging outcomes of such interventions.<sup>132</sup>

### Diverting people who use drugs from the criminal justice system

Diversion programmes are generally conducted by police and/or require police input. Diversion strategies examined in this section focus on pre-arrest and pre-court strategies. Police are most involved in pre-arrest and pre-trial diversion strategies, although they can have a role in other forms of diversion.<sup>133</sup>

### Why divert offenders?

Diversion or referral programmes aim to prevent first offenders who are unlikely to re-offend from entering the criminal justice system. Once someone has entered the criminal justice system, the effects of incarceration and of having a criminal record can be far-reaching and generate long-lasting impediments. Offenders who have previously been apprehended can also benefit from referral, if encouraged to enter a treatment programme in order to reduce their illicit drug use and associated criminal activity.

*Pre-arrest diversion* can occur when police notice a minor offence (such as the simple non-violent possession of illicit drugs) but have not made an arrest. Methods range from informal warnings through to deferred sentencing.

Verbal warnings can involve taking the person home or instructing them to move from the place where the problem occurred. Warnings present an excellent opportunity to provide information regarding MAT and other drug treatment and harm reduction material. Cautioning schemes can include referral to education and/or drug treatment (including MAT). The process generally involves following set procedures and some record keeping. For example, a 'cannabis cautioning' scheme is where a person who is apprehended for the possession or use of a small amount of cannabis is not charged but is instead given a 'caution notice' with some accompanying health information. At police discretion, the offender is offered the option of a caution, provided that the offender admits to the offence. A person can accumulate two cautions only, and a subsequent offence will result in prosecution. Usually in such a scheme the person's name is recorded for future reference, but the person has been diverted from the criminal justice system. Cannabis educational information and a referral for a cannabis education session accompany such a caution.<sup>134 135</sup>

A working example can be found in the Cannabis Cautioning Scheme of New South Wales (NSW) which provides for formal cautioning of adult offenders detected for minor cannabis offences, with the aim of using police intervention to assist offenders to consider the legal and health ramifications of their cannabis use and seek treatment and support. Under this scheme, adults detected by police using or in possession of not more than 15 grams of dried cannabis and/or equipment for using the cannabis may receive a formal police caution, rather than face criminal charges and court proceedings. The Scheme only applies to adults and allows police to exercise their discretion in appropriate cases and issue a caution. Police are still able to decide to formally charge offenders. A person can only be cautioned twice and cannot be cautioned at all if they have prior convictions for drug offences or offences involving violence or sexual assault. The Scheme also does not apply to those caught supplying cannabis, and drug dealers continue to be arrested and prosecuted. The formal police caution warns of the health and legal consequences of cannabis use and provides each cautioned person with information about treatment and support services. In particular, the caution notice advises offenders that they can call the Alcohol and Drug Information Service (ADIS) for confidential help and information regarding their cannabis use. Persons who receive a second, final, caution are required to contact the ADIS for a mandatory education session about their cannabis use.<sup>136</sup>

### C. Post arrest referral of non-violent drug offenders to drug treatment

Referral after arrest (including drug courts, bail conditions and deferred sentencing) also targets non-violent illicit drug users, aiming to redirect them into non-incarceration alternatives such as treatment. There must usually be sufficient admissible evidence that the offender is using or in possession of a small (i.e. non-trafficable) quantity of illicit drugs, and that the drugs must be for personal use only. In the United Kingdom, an example is found in the national Drug Interventions Programme, which seeks to address the problem of drug-related offending by encouraging drug using offenders to access treatment (including MAT). In high crime areas in England and Wales, people who are arrested for certain drug related offences are tested upon arrest or charge for illicit drug use. If such individuals test positive, an assessment of their drug use is undertaken which may then result in referral to treatment. Most of those referred into drug dependence treatment demonstrate reduced involvement in property crime. In lower crime areas, arrest referral programmes without drug testing are used in order to divert people who use drugs into appropriate treatment. Evaluations of arrest referral programmes indicate post-arrest reductions in drug use.<sup>137</sup>

An ongoing law enforcement operation in Brighton (United Kingdom), named 'Operation Reduction,' has been subject to an independent evaluation. Law enforcement personnel, working alongside treatment providers, offer people using drugs who are funding their drug use

via drug sales a fast-track route into treatment. The evaluation of this operation found it was well received by stakeholders, with reductions in criminal activity at both district and individual levels. Cost benefits were also demonstrated, with estimates that for every £1 spent on the operation, £3 was saved on crime costs.<sup>138</sup>

These examples show how effective the development of partnership approaches between local level law enforcement agencies and community service providers (including MAT clinics) can be, in terms of crime reduction, cost-effectiveness and positive public health outcomes.

**Drug courts** are alternative courts that handle the cases of non-violent drug using offenders under the justice system, generally diverting offenders towards assistance, including MAT. Drug courts involve intersectoral collaboration where judiciary, law enforcement, social services, and treatment communities work together to assist non-violent offenders outside a prison context. The results are very encouraging. Graduates of drug courts are, according to current evaluations, less likely to be rearrested than persons processed through traditional courts. Findings from drug court evaluations show that participation in drug courts results in fewer rearrests and reconvictions, or longer periods between arrests.<sup>139</sup>

In many jurisdictions, the trend towards the use of drug courts emerged from the reality that addressing illicit drug use through law enforcement mechanisms would continue to pose disproportionate challenges for the criminal court system. For example, in the US in 2004, 53% of persons in state prison were identified with a drug dependence or abuse problem, but only 15% were receiving professional treatment.<sup>140</sup> Drug-related crime continues to present a burden to society, one that punitive 'zero tolerance' enforcement efforts alone have failed to curb.

The role of law enforcement in drug courts is significant. A survey of US law enforcement officers involved in drug court work was conducted in 2000, in which respondents outlined their potential duties to include:

- discussing possible drug court participation with offenders at the time of their arrest;
- referring cases of potentially eligible defendants to drug court staff;
- attending drug court hearings;
- assisting with participant supervision;
- serving on drug court oversight committees; and
- attending drug court graduations.<sup>141</sup>

The same survey asked law enforcement respondents to indicate the impact that the drug court has had upon police capability to respond to criminal activity or otherwise carry out functions. The most frequently cited impact related to the new relationships that the programme had generated between law enforcement and other justice system and community agencies, including greater interaction with local substance dependency treatment service providers and community groups. Other frequently cited impacts related to the availability of a more effective response to arrests of illicit drug users.<sup>142</sup>

Some drug court programmes include referral for assessment and treatment as a condition of bail. For example, in the State of Victoria, Australia, police run the 'Court Referral Education, Drug Intervention and Treatment (CREDIT)' programme. The CREDIT programme can be offered to offenders with substance use issues as part of bail proceedings after initial arrest.

Persons charged with any offence who have an immediately presenting drug problem are referred by police for assessment by a drug clinician based at the court and, where appropriate, the alleged offender is diverted into a recommended treatment regime by the magistrate as a condition of bail. However, this option is only available at Magistrates' Courts where there is a court-appointed drug clinician.<sup>143</sup>

A report on improving the quality of drug courts found that drug court programmes should be flexible enough to allow the appropriate use of the range of medications that have been shown to be useful in the treatment of opioid dependence. MAT medications such as methadone and buprenorphine have been clearly shown to improve treatment outcomes for drug dependent individuals, and the data fully justify the conclusion that MAT should be considered as an integral part of any drug court treatment programme. The authors found that "... to deny drug court participants the option of receiving medications for their treatment is in our opinion unethical."<sup>144</sup>

Further, the cost-effectiveness of the use of medications in preventing re-incarceration more than offsets the additional costs of providing medications. Evaluations of the net costs and benefits of drug courts across the US show that drug courts save money compared to simple probation and/or incarceration, primarily due to reductions in arrests, case processing, jail occupancy, and victimisation costs.<sup>145</sup> <sup>146</sup>

Another potential diversion method is via **deferred sentencing**, which targets persons who are drug dependent and have been found guilty of an offence. Sentencing can be deferred for a set period with a specific condition to attend drug treatment. Pre-sentence clinical drug assessments are undertaken and a treatment plan is recommended to the court. Offenders then attend the prescribed drug treatment and a report on progress is made to the court prior to sentencing.<sup>147</sup>

### Guidance points

Drug control regulators within home affairs or public security ministries can springboard reforms to establish a better balance between law enforcement concerns and more streamlined access to opioid medicines for those in need.

Arrest and court referral schemes are effective in moving dependent drug users towards social reintegration. Law enforcement agencies should put greater emphasis on referral to treatment, rather than on the more resource-intensive process of prosecution and imprisonment.

Three key approaches can be used by law enforcement to enhance access to MAT:

- 1. referring informally or formally encouraging people who use drugs to enter treatment through the provision of advice;
- 2. more formal diversion from the criminal justice system (including caution schemes and drug courts); and
- 3. use of discretion in policing around MAT dispensing sites.<sup>148</sup>

In order to effectively implement such reforms, changes in police attitude and practice may be required. These can be shaped by an awareness of the benefits of MAT, including those of relevance to law enforcement (i.e. those related to cost-effectiveness and reduced criminal

activity) and a working understanding of methods which police can employ to facilitate access to MAT.

This section examining referral and diversion into treatment (including MAT) highlights that law enforcement plays a vital role, expressed through regulatory reform, developing partnerships with community health providers, the balanced application of police discretion, and a range of formal and informal diversion methods. Through knowledge of the key concepts, experience and evidence detailed in this guidance tool and accompanying training, law enforcement implementers will understand the rationale and methods needed to ensure more efficient policing of, as well as more streamlined access to, essential medical opioids and MAT.

# 5. Assisting ongoing treatment for people in, and released from, custodial settings

In principle, the same standard and range of medical care available to the community should be available to detainees and prison inmates, whether in custody for a matter of days or years. Where any barriers to continued access to legitimate prescribed medications – including medical opioids and MAT – are identified, they should be relaxed or removed. For prisons in many jurisdictions it is strongly recommended that MAT programmes be introduced for drug dependent individuals. The case for integrating MAT programmes within short and longer term detention and penitentiary facilities is compelling. Drug regulators, police and corrections officers can play an important role in ensuring continuity of access to prescription and MAT medications between community and prisons.

### A The need for treatment continuity

It is increasingly acknowledged that the treatment of people receiving pain relief or MAT should be continued if such individuals are moved from the community to prison (i.e. incarcerated), and from prison to the community (i.e. upon release). MAT in prison has been shown to result in reduced non-medical drug use, reduced transmission of blood borne viruses, reduced mortality and reduced criminal activity after release.<sup>149</sup> Prisoners receiving MAT are, upon release, more likely to enter into and remain enrolled in drug treatment programmes, and less likely to experience drug overdose.<sup>150</sup> This requires both linkage between relevant sectors, such as law enforcement, corrections and health, and the establishment of MAT and pain relief and palliative care programmes within prisons, comparable to such programmes in the community.<sup>151 152</sup>

Continuity of access to all medical opioids is clearly supported when moving into, or out of, custody. For example, Australian national guidelines state that; "Every prisoner is to have access to evidence-based health services provided by a competent, registered health professional who will provide a standard of health services comparable to that of the general community."<sup>153</sup> The WHO also recommends that patients of MAT taken into custody should be able to continue such treatment.<sup>154</sup> Jurisdictions in which MAT is available in the community are advised to urgently implement such programmes in custodial settings.<sup>155</sup>

Continuity in treatment for pain relief should never be impeded – the alternative is both unethical and inhumane. Continuity of treatment for people on MAT is similarly critical. Without prompt attention, a person on MAT can experience withdrawal symptoms, presenting additional challenges for custodial officers. Studies have shown that people who discontinue MAT upon imprisonment tend to continue to use drugs chaotically both during and after prison terms. Drug overdose post-release from prison is common, partly because of reduced tolerance to pre-imprisonment drug doses. This highlights the necessity for linkages between prisons and community providers. Overdose risk is dramatically reduced where the drug dependent prisoner receiving MAT in prison is provided with an uninterrupted transition to community prescription, post-release. Continuity with MAT access in prisons is also an indicator of retention, with high proportions of those receiving MAT in prison arrest, and thereafter in prison, undermines the merits of MAT prescribing along with the benefits to individuals and to the community.<sup>156</sup>

## **B** Common concerns among law enforcement and corrections personnel related to MAT in prisons.

Some police and custodial officers share common concerns and misunderstandings about drug dependency and MAT. Below is a list of conflicting perceptions related to MAT in prisons, followed by evidence-based responses.

### 'Prisoners should abstain from drug use while in custody'

While this may be a well intentioned notion, it has proved unachievable in prison systems. Drug use and sexual activity continue in prisons, and there are documented cases of HIV infection within such settings.<sup>157</sup> While inmates are generally able to access drugs which they can inject, restrictive policy means that sterile injecting equipment is less available, and that sharing of non-sterile injecting equipment is the norm. One study reported that 50% of inmates were drug injectors, that almost half injected while in prison and that 94% shared injecting equipment.<sup>158</sup> These circumstances, known to lead to HIV outbreaks within prisons,<sup>159</sup> are still common in many jurisdictions, and particularly pronounced in developing and transitional countries. Detainees return home with the attendant risk of transmission to partners, families and communities.<sup>160</sup> MAT has been shown to reduce illicit drug use activity both within and outside prisons systems.

### 'MAT just substitutes one drug for another'

In fact, the pharmacological properties of heroin are very different to that of medication assisted treatment. MAT uses long-acting opioids such as methadone and buprenorphine which – if provided for a sufficient (sometimes lengthy) period and at a sufficient dose – can stabilise people who are dependent upon heroin and other short-acting opioids. With a dose per day, MAT prevents withdrawal in drug dependent individuals so that they can function normally in prison or in the community. Research has shown that treatment for drug dependent offenders during and after incarceration can have a significant beneficial effect upon future drug use, criminal behavior and social functioning. "Combining prison and community-based treatment for drug addicted offenders reduces the risk of both recidivism to drug-related criminal behavior and relapse to drug use."<sup>161</sup> The many health and other benefits associated with medication-assisted treatment, including reduced criminal activity and reduced illicit drug use, makes MAT a multipurpose tool in prisoner care and management (*see also Section 1D on MAT*).

### 'Drug users are just weak willed'

The WHO and the United Nations Office on Drugs and Crime (UNODC) recognise that drug dependence is a serious health condition. "Drug dependence is considered a multi-factorial health disorder that often follows the course of a relapsing and remitting chronic disease."<sup>162</sup> Withdrawal or detoxification in dependent drug users is ordinarily followed by repeated relapse. Some drug dependency experts liken heroin dependency to diabetes – to withhold insulin from the diabetic is akin to withholding MAT from a dependent person. MAT has proven the most effective response to illicit heroin injecting in terms of both public health and law enforcement imperatives (*see also Section 1D on heroin dependence*).

### 'MAT doesn't improve anything for me'

Detainee and prisoner case management can be safer and more straightforward when not dealing with individuals undergoing withdrawal or engaging in risky drug taking in prison settings. Police and custodial officers report significant benefits associated with MAT provision in custodial settings. The common experience is that:

- MAT has a positive effect on institutional behaviour by reducing drug-seeking behaviour and improving prison safety;
- a range of common initial concerns among prison personnel (such as increased violence or drug diversion) have not been realised when MAT is actually implemented in prisons;
- prisoners and prison personnel report that MAT has a beneficial impact on the prison culture.<sup>163</sup> <sup>164</sup>

### 'MAT doesn't improve anything for inmates'

The provision of MAT in prison settings has been associated with the following benefits for prisoners:

- reduced injection of illicit drugs;
- reduced HIV and hepatitis infection;
- reduced mortality;
- lower rates of post-release drug use among MAT clients;
- reduction in the use of non-sterile injecting equipment;
- increase in the use of condoms in sexual relationships;
- reduction in the number of overdoses;
- MAT can improve the delivery of ART to HIV-positive drug dependent prisoners.

Furthermore – and of benefit to MAT clients, law enforcement and the community at large – MAT has a positive effect upon criminal recidivism and re-incarceration, particularly if methadone is provided for longer, uninterrupted periods and if moderate-to-high doses of methadone are provided. It has also been found that re-incarceration is less likely among those prisoners who receive adequate MAT while incarcerated.<sup>165 166 167</sup>

Therefore, prison systems in jurisdictions where MAT programmes are available in the community are urged to introduce and scale up MAT for detainees and prisoners.

### *C* Examples of good practice from Iran, India, Poland, USA, and Australia

Good practice examples of MAT and other medication continuity when moving into custody, or upon release, are available across a range of socioeconomic environments including in resource-poor countries.<sup>168</sup>

In Iran, where MAT has been available through 'triangular clinics' in prisons since 2003, methadone treatment is one component of broader HIV prevention efforts. At the end of 2006 there were 55 'triangular clinics' in prisons in Iran covering 33% of prisoners. Further, there were another 34 such clinics located in after-care centres in the community. By the beginning of 2007, the clinics were providing MAT (in the form of methadone maintenance therapy) for 55% of prisoners in need, with plans to increase coverage to 80-99% by 2008.<sup>169</sup>

In Kolkata, law enforcement is confronted with the problem of people who use drugs going into withdrawal while in custody. As law enforcement personnel are not trained to adequately respond to such situations, working relationships with NGOs are becoming increasingly

important. NGO personnel are permitted to enter the prison and deliver harm reduction information and advice. Upon release, prisoners are encouraged to access and utilise drop-in centre facilities and services, including both NSP and MAT programmes. Kolkata law enforcement authorities have expressed their support for harm reduction programmes – including MAT – to be available in prisons in the near future.<sup>170</sup>

Prisons in Poland allow individuals who have been receiving MAT in the community to continue such treatment while incarcerated.<sup>171</sup> MAT is also available in prisons in Australia, Canada and Puerto Rico, but remains of limited accessibility in the USA.<sup>172</sup>

However, even in the USA, trial methadone programmes treated inmates and over time demonstrated statistically significant differences in decreased criminal recidivism among those who received MAT in prison. A prison methadone programme running at Rikers Island from 2000, established the 'model'. A number of correctional facilities have since indicated an interest in using MAT to treat chronic heroin dependence, based on the success of the Rikers Island model. The Rikers Island experience indicates that providing access to such medication assisted treatment in correctional facilities is an extremely effective method of reducing recidivism.<sup>173</sup>

In some jurisdictions, case management models for systemising ongoing access to medical opioids and MAT in prison have emerged. For the many people who enter custody with an alcohol or other drug dependency, incarceration may result in unintended detoxification, which also presents additional challenges for custodial officers. In Victoria, Australia, the Custodial Risk Management Unit provides a collaborative model that could be applied in other jurisdictions. The Unit was established in recognition of the health needs of people in police custody, and to make the transition from the community into custody safe. It works proactively with Victoria Police to improve health outcomes and reduce risks for people in police care. It is staffed by a doctor and a team of nurses, and supported by a network of general practitioners. Custodial nurses offer a comprehensive health assessment to everyone held in custody, and if necessary a care plan is developed in consultation with a medical officer and the police. The doctors prescribe medication when it is required to continue regular treatment and may initiate treatment for alcohol or drug withdrawal. The nurses help people find pharmacotherapy prescribers, pharmacies and other relevant services. If the person is going to prison, referral is made to the appropriate services in the jail.<sup>174</sup>

The custodial nursing service has been well supported by police. Since the commencement of the current programme in 2002, police report that there has been a dramatic reduction in the number of health and welfare issues in custody. The balance between the health of people in custody and the workings of the justice system can present challenges which the Unit manages through liaison with external agencies to ensure treatment continuation or assessment with potential for MAT initiation. The major success of this unit has been the collaboration between health professionals and the police. By working together in police stations and being involved in the Unit's functions, police have come to recognise some of the complexity of the health needs of the people in their care, and the benefits to custodial management of rapid access to ongoing medical treatment for prisoners. The result has been better care for all people in custody.<sup>175</sup>

Also by the Victorian Police in Australia, the *Victorian Prison Opioid Substitution Therapy Program: Clinical and Operational Policy and Procedures* document details instructions for the development of MAT in prisons, along with the aims, objectives and underlying principles for providing MAT in Victorian prisons. The document details practical, programmatic components including maintenance and induction programmes, clinical monitoring and review, service delivery arrangements, release referral and the monitoring and evaluation processes.<sup>176</sup>

The corresponding policy document of the Corrective Services Department of the Queensland state government states that:

"Opioid pharmacotherapy maintenance treatment will be available for remandees and offenders serving sentences of less than 12 months where they were undertaking community opioid replacement therapy at the time of reception to custody. Pregnant female (sentenced) offenders in custody who do not meet the criteria above may undertake opioid replacement treatment for the duration of their pregnancy. Partnership arrangements will be developed with Queensland Health to enable continuation of opioid maintenance pharmacotherapy for offenders following their release."<sup>177</sup>

An individual taken into law enforcement custody should "be afforded the opportunity to receive their regular dose from an authorised prescriber or dispenser," and the custodial officer "should get the name of the detained person's medical practitioner or dosing point and call them as soon as possible," in order to avoid any unnecessary delay in dosing.<sup>178</sup>

Generally, the responsibility for making alternative dosing arrangements for MAT clients taken into custody lies with the prescriber / dispenser. "If a detained person has any take-away doses of methadone on them, these should be taken into possession and accounted for as prisoner property in the usual way. The doses should be recorded as part of the person's property (including whether the bottle/s were full or empty). If the person is in possession of an illegally obtained dose of methadone, police can contact the prescriber (whose name will be on the label on the bottle containing the dose) and tell the medical practitioner what has been found. The name of the person to who the dose was prescribed should also be reported to the prescriber. This allows the prescriber to be made aware that one of his or her clients may have been trading, rather than taking, the daily dose."<sup>179</sup> Other prescription and dosing guidelines provided in the community can be adapted for use in prisons. For example, to minimise diversion, it is advised that dosing take place in view of the dispenser.

Research from the US indicates that under-treated pain is common among prisoners. Further, analysis has shown that survival for cancer patients in prison was substantially inferior to that of a non-incarcerated.<sup>180</sup> Common barriers identified through interviews with practitioners and prisoners included concern over drug misuse / diversion, systemic obstacles and lack of prisoner credibility. The power imbalance experienced between prisoners and prison officers is rarely conducive to open and honest communication. The authors suggest that one solution to evident trust issues could be the use of written contracts between inmates, practitioners and prison authorities. In addition, the formation of a panel involving pain experts, drug abuse experts, prison authorities, and inmate advocates could examine disputed cases of drug diversion in order to prevent potential discontinuation of pain medication. Prison security protocols which restrict pain management could also be reduced through a multidisciplinary approach involving

prison authorities and medical personnel. Educational sessions on pain assessment and management can address the lack of capacity for primary care practitioners.<sup>181</sup>

### Guidance points

The case for integrating MAT programmes within short and longer term detention and penitentiary facilities is compelling. In providing MAT and other prescribing services in prisons, law enforcement and corrections personnel are able to minimise harmful drug use practices and drug diversion in prison environments, and ensure continuity in treatment if the detainee / inmate is currently prescribed medical opioids or MAT. In these ways, law enforcement can bring better balance to considerations of public health, public security, human rights, and development.

Where any barriers to continued access to legitimate prescribed medications, including medical opioids and MAT are identified, they should be relaxed or removed. The WHO recommendations regarding MAT in prisons calls upon authorities in countries where MAT is available in the community to urgently introduce MAT programmes within prisons, and to expand implementation to scale as soon as possible. "Particular efforts should be undertaken to ensure that prisoners on MAT prior to imprisonment are able to continue this treatment upon imprisonment, without interruption."<sup>182</sup>

### 6. Conclusion

The experience and guidance presented above is not exhaustive, but provides the context for law enforcement measures towards improved legitimate access to medical opioids and MAT.

In order to successfully establish training and guidance for law enforcement regarding access to essential medical opioids, policy must match practice at all levels. Where policies are overly restrictive, or hindering access to medical opioids, policy reform is required. Regulatory review as well as policing of medical opioids and MAT supplies must be shaped by the need to ensure sufficient and uninterrupted supply, as well as drug control objectives. There has been a global tendency towards emphasis on attempts to eliminate illicit drug use and the diversion of medical opioids (albeit with little or no success) by law enforcement, at the unacceptable expense of millions of people who require, but are denied, pain relief and/or MAT. Excessively restrictive drug control regulations or enforcement practices do interrupt or limit the supply of medical opioids and MAT to end users.

Fortunately, without altering the stipulations of the UN Drug Conventions, much can be done to rectify the imbalance. Governments have substantial flexibility in how to manage the medical opioid and MAT supply chain to satisfy UN Drug Convention requirements. Cultivation and manufacture require application, central oversight and licensing, while storage and transport must be under 'legal authority' of some description. While governments may, under the Convention, impose additional requirements if deemed necessary – such as requiring that all prescriptions be written on government forms – this is an option governments can adopt or ignore according to national needs. As WHO has observed, " ... this right must be continually balanced against the responsibility to ensure opioid availability for medical purposes."<sup>183</sup> Any regulations that unnecessarily impede access to medical opioids and MAT are actually inconsistent with both the UN Drug Conventions and the Universal Declaration of Human Rights, which require countries to achieve balance between legitimate availability and preventing illicit activity. The UK opioid transport experience provides an interesting example of streamlining access to medical opioids by simplifying regulations.

To combat misplaced stigma attached to the prescription and use of medical opioids, it is important that regulators and law enforcement understand the need for medical opioids, including MAT, and take a more balanced approach in public messages to physicians and how they handle routine investigations of medical practice.<sup>184</sup>

In order for law enforcement to operationalise reforms towards appropriate access to medical opioids and MAT, some changes in practice and attitude may be necessary. For example, diversion prevention methods must be proportionate and take into account that medical opioids and MAT are essential medications, and that while potential diversion should be identified and averted, the potential illicit activities of a few cannot be allowed to interrupt or delay sorely needed medical provisions for many. It should be recalled that, at the international level, diversion of opioid medications from the licit trade into illicit channels remains relatively rare and in small quantities compared to the necessarily large trade flow.<sup>185</sup> Diversion control should be based on evidence to better identify and respond to key diversion sources. Responses for medical opioid diversion prevention must be carefully considered and balanced against potential unintended consequences. Adjustments may be necessary to better ensure the required flow of
licit opioid medications to end users, as well as disrupting any significant diversion sources. Law enforcement must adopt appropriate access to medical opioids and MAT as a core goal alongside control of illicit use. Working partnerships between law enforcement and the health sector, including MAT providers, will also be required to correct the imbalance.

## 7. DCAM website

Those who wish to understand more about improving access to medical opioids are invited to visit the Drug Control and Access to Medicines (DCAM) Consortium website: <u>http://www.dcamconsortium.net/</u>

The DCAM Consortium has organized an International Coordinating Committee, bringing together a global group of representatives from palliative care, drug dependency treatment, global health, global drug policy, and other organizations to work together to accelerate policy change. The DCAM website is designed to assist all organizations and individuals interested in accelerating policy change to access the data and guidance materials they need.

The DCAM Consortium website provides a short statement on the problem (with a map of global opioid consumption) and the DCAM global and national level solutions, involving a systematic programme of assessment, planning, coordination, and intervention implementation.

The website presents the DCAM *Compendium of INCB Statements on Access to Medicines*, bringing together INCB statements on a range of specific issues related to the need for medical and scientific access to medical opioids and MAT.

A range of resources with direct links is detailed within the DCAM website. These resources are searchable by keyword and by country.

## 8. Annexes

## Key documents:

Burris, S. & Davis, C.S. *A Blueprint for Reforming Access to Opioid Medications* <u>http://ssrn.com/abstract=1356815</u>

Closing the Gap: Case Studies of Opioid Access Reform in China, India, Romania & Vietnam <u>http://ssrn.com/abstract=1356769</u>

*Review of Global Policy Architecture and Country Level Practice on HIV/AIDS and Drug Treatment* <u>http://ssrn.com/abstract=1357336</u>

WHO Policy Brief on Effectiveness of Drug Dependence Treatment in Preventing HIV Among Injecting Drug Users http://www.who.int/hiv/pub/idu/e4a-drug/en/

WHO Evidence for action technical paper: Interventions to address HIV in prisons: Effectiveness of interventions to address HIV in prisons <a href="http://www.who.int/hiv/pub/idu/prisons\_effective/en/index.html">http://www.who.int/hiv/pub/idu/prisons\_effective/en/index.html</a>

## References

<sup>1</sup> United Nations Commission on Narcotic Drugs Report on the Fifty-Third Session (2 December 2009 and 8-12 March 2010). Economic and Social Council Official Records, Supplement No. 8. 2010.

<sup>2</sup> World Health Organization (WHO). Narcotic and Psychotropic Drugs – Achieving Balance in National Opioids Control Policy: Guidelines for Assessment. WHO: Geneva. 2000. In Burris, S. & Davis, C.S. A Blueprint for Reforming Access to Opioid Medications: Entry Points for International Action to Remove the Policy Barriers to Care and Treatment. Temple University, Beasley School of Law, Center for Health Law, Policy and Practice: Philadelphia. 2009.

<sup>3</sup> Burris, S. & Davis, C.S. A Blueprint for Reforming Access to Opioid Medications: Entry Points for International Action to Remove the Policy Barriers to Care and Treatment. Temple University, Beasley School of Law, Center for Health Law, Policy and Practice: Philadelphia. 2009.

<sup>4</sup> International Narcotics Control Board (INCB). *Report of the International Narcotics Control Board for 1995: Availability of opiates for medical needs.* United Nations; New York, NY. 1996.

<sup>5</sup> World Health Organization (WHO). Narcotic and Psychotropic Drugs – Achieving Balance in National Opioids Control Policy: Guidelines for Assessment. WHO: Geneva. 2000.

<sup>6</sup> Burris, S. & Davis, C.S. *Blueprint for Reforming Access to Therapeutic Opioids: Entry Points for International Action to Remove the Policy Barriers to Care.* Centers for Law and the Public's Health: A Collaborative at the Johns Hopkins and Georgetown Universities. 2008.

<sup>7</sup> Human Rights Watch. *Lessons Not Learned: Human Rights Abuses in the Russian Federation*. New York: Human Rights Watch. Vol. 16:5D. 2004.

<sup>8</sup> Open Society Institute (OSI). Drugs, AIDS and Harm Reduction. OSI: New York. 2001.

<sup>9</sup> Human Rights Watch. *Rhetoric and Risk: Human Rights Abuses Impeding Ukraine's Fight Against HIV/AIDS* New York: Human Rights Watch. Vol. 18:2D. 2006.

<sup>10</sup> Wolfe, D. & Malinowska-Sempruch, K. Seeing Double: Mapping Contradictions in HIV Prevention and Illicit Drug Policy Worldwide. In Beyer, C. (editor). Public Health and Human Rights: Evidence Based Approaches. 2007.

<sup>11</sup> Burris, S., Blankenship, K., Donoghoe, M. et al. 'Addressing the "Risk Environment" for Injection Drug Users: The Mysterious Case of the Missing Cop'. *The Milbank Quarterly* 82, No. 1:125. 2004.

<sup>12</sup> Kerr, T., Small, W. & Wood, E. 'The Public Health and Social Impacts of Drug Market Enforcement: A Review of the Evidence'. *International Journal of Drug Policy* 16: 210. 2005.

<sup>13</sup> Rhodes, T., Platt, L., Sarang, A., et al. 'Street Policing, Injecting Drug Use and Harm Reduction in a Russian City: A Qualitative Study of Police Perspectives'. *Journal of Urban Health: Bulletin of the New York Academy of Medicine* 83, No. 5: 911. 2006.

<sup>14</sup> Csete, J. *Do Not Cross: Policing and HIV Risk Faced by People Who Use Drugs*. Canadian HIV/AIDS Legal Network: Toronto. 2007.

<sup>15</sup> Preventing the transmission of HIV among drug abusers: A position paper of the United Nations System. Annex to the Report of 8th Session of Administrative Committee on Co-ordination Subcommittee on Drug Control. UNAIDS: Geneva. 28-29 September 2000.

<sup>16</sup> UNODC/WHO/UNAIDS. *HIV Prevention, Care, Treatment and Support in Prison Settings: A Framework for an Effective National Response.* UNODC/WHO/UNAIDS: Geneva. 2006.

<sup>17</sup> World Health Organization (WHO). *Guidelines on HIV Infection and AIDS in Prison*. WHO: Geneva. 2007.

<sup>18</sup> World Health Organization (WHO). *The ICD-10 Classification of Mental and Behavioural Disorders: Clinical descriptions and diagnostic guidelines.* WHO: Geneva. 1992.

<sup>19</sup> Anthony, J., Warner, L. & Kessler, R. 'Comparative epidemiology of dependence on tobacco, alcohol, controlled substance and inhalants: basic findings from the National Comorbidity Survey'. *Exp Clin Psychopharmacol* 2:244-268. 1994.

<sup>20</sup> Hojsted, J. & Sjogren, P. 'Addiction to opioids in chronic pain patients: A literature review'. *European Journal of Pain* 11:490-518. 2007.

<sup>21</sup> Fishbain, D.A., Cole, B., Lewis, J., Rosomoff, H.L. & Rosomoff, R.S. 'What percentage of chronic nonmalignant pain patients exposed to chronic opioid analgesic therapy develop abuse/addiction and/or aberrant drug-related behaviors? A structured evidence-based review'. *Pain Medicine* 9(4): 444-59. 2008.

<sup>22</sup> Littlejohn, C., Baldacchino, A. & Bannister, J. 'Chronic non-cancer pain and opioid dependence'. *J R Soc Med* 97(2): 62–65. 2004.

<sup>23</sup> Portenoy, R.K. *Opioid therapy for nonmalignant pain: current status.* In Fields, H.L. & Liebeskind, J.C. (editors). *Pharmacological approaches to the treatment of chronic pain: new concepts and critical issues: the Bristol-Meyers Squibb Symposium on Pain Research. Progress in pain research and management.* Vol. 1: 247-87. IASP Press: Seattle. 1994.

<sup>24</sup> Burris, S. & Davis, C.S. *Blueprint for Reforming Access to Therapeutic Opioids: Entry Points for International Action to Remove the Policy Barriers to Care.* Centers for Law and the Public's Health: A Collaborative at the Johns Hopkins and Georgetown Universities. 2008.

<sup>25</sup> World Health Organization (WHO). *Narcotic and Psychotropic Drugs – Achieving Balance in National Opioids Control Policy: Guidelines for Assessment*. WHO: Geneva. 2000.

<sup>26</sup> Gerra, G., Maremmani, I., Capovani, B., Somaini, L., Berterame, S. & Tomas, J. 'Long-Acting Opioid-Agonists in the Treatment of Heroin Addiction: Why Should We Call Them "Substitution"?' *Substance Use & Misuse* 44:663-671. 2009.

<sup>27</sup> Burris, S. & Davis, C.S. *Blueprint for Reforming Access to Therapeutic Opioids: Entry Points for International Action to Remove the Policy Barriers to Care.* Centers for Law and the Public's Health: A Collaborative at the Johns Hopkins and Georgetown Universities. 2008.

<sup>28</sup> New South Wales Police. *Methadone and other Pharmacotherapies: Guidelines for Police*. NSW Police: Sydney. 2003.

<sup>29</sup> American Association for the Treatment of Opioid Dependence (AATOD). *Fact Sheet for Drug Courts: Methadone Maintenance and other Pharmacotherapeutic Interventions in the Treatment of Opioid Dependence.* AATOD: New York. 2004. http://www.aatod.org/fact\_drug\_court.html

<sup>30</sup> New South Wales Police. *Methadone and other Pharmacotherapies: Guidelines for Police*. NSW Police: Sydney. 2003.

<sup>31</sup> World Health Organization (WHO). *List of Essential Medicines*. (15<sup>th</sup> List). 2007. http://www.who.int/medicines/publications/08\_ENGLISH\_indexFINAL\_EML15.pdf <sup>32</sup> International Narcotics Control Board (INCB). *Report of the International Narcotics Control Board for 2003*. United Nations; New York, NY. 2003.

<sup>33</sup> United Nations. *Single Convention on Narcotic Drugs, 1961*. United Nations: New York. 1961.

<sup>34</sup> United Nations. *Convention on Psychotropic Substances, 1971.* United Nations: New York. 1971.

<sup>35</sup> Drug Control and Access to Medicines (DCAM) Consortium. DCAM in Central Asia: Project Description. 2009.

<sup>36</sup> Joranson, D.E., Ryan, K.M. & Maurer, M.A. *Opioid Policy, Availability, and Access in Developing and Nonindustrialized Countries.* In Fishman, S.M., Ballantyne, J.C. & Rathmell, J.P. (editors). *Bonica's Management of Pain. (Fourth edition).* Lippincott Williams & Wilkins: Baltimore, MD. 2010: 194-208. http://www.painpolicy.wisc.edu/publicat/10bonica/Joranson\_2010\_Bonica%20Chapter%2016.pdf

<sup>37</sup> Drug Control and Access to Medicines (DCAM) Consortium. *Communicating Drug Policy: Report of an International Consultation on Ways Forward for Licit Opioid Access as part of Drug Control.* 2009.

<sup>38</sup> Drug Control and Access to Medicines (DCAM) Consortium. DCAM in Central Asia: Project Description. 2009.

<sup>39</sup> Ali, R., Burrows, D., Gowing, L., Vial, R., Walsh, N., Onyancha, P. & Finnerty, E. Architecture and Country Level Practice on HIV/AIDS and Drug Treatment. AIDS Project Management Group (APMG): Sydney. 2008.

<sup>40</sup> Cook, C. & Kanaef, N. *The Global State of Harm Reduction 2008: Mapping the response to drug-related HIV and hepatitis C epidemics*. International Harm Reduction Association (IHRA): London. 2008.

<sup>41</sup> Ali, R., Burrows, D., Gowing, L., Vial, R., Walsh, N., Onyancha, P. & Finnerty, E. Architecture and Country Level Practice on HIV/AIDS and Drug Treatment. AIDS Project Management Group (APMG): Sydney. 2008.

<sup>42</sup> Burris, S. & Davis, C.S. *Blueprint for Reforming Access to Therapeutic Opioids: Entry Points for International Action to Remove the Policy Barriers to Care.* Centers for Law and the Public's Health: A Collaborative at the Johns Hopkins and Georgetown Universities. 2008.

<sup>43</sup> Gallagher, R. 'Opioids in chronic pain management: Navigating the clinical and regulatory challenges'. *The Journal of Family Practice Supplement* 53, 10: S23-S32. 2004.

<sup>44</sup> Hoffmann, D.E. & Tarzian, A. J. 'Achieving the right balance in oversight of physician opioid prescribing for pain: The role of state medical boards'. *Journal of Law, Medicine & Ethics* 31, 1: 21-40. 2003.

<sup>45</sup> Ziegler, S.J. & Lovrich, N.P. 'Pain relief, prescription drugs, and prosecution: a four-state survey of chief prosecutors'. *Journal of Law, Medicine & Ethics* 31, 1: 75-100. 2003.

<sup>46</sup> Aitken, C., Moore, D., Higgs, P., Kelsall, J. & Kerger, M. The impact of a police crackdown on a street drug scene: evidence from the street. *International Journal of Drug Policy*, Volume 13, Issue 3: 193-198. 2001.

<sup>47</sup> Burris, S., Blankenship, K., Donoghoe, M. et al. 'Addressing the "Risk Environment" for Injection Drug Users: The Mysterious Case of the Missing Cop'. *The Milbank Quarterly* 82, No. 1:125. 2004.

<sup>48</sup> Calvani, S. & Bazant, W. *Building synergies between public health and public security for the prevention of drug related harm*. Presentation at ICAA Conference in Toronto. 2003.

<sup>49</sup> United Nations. 1961 Single Convention on Narcotic Drugs, Part 1: The International Control System for Narcotic Drugs. United Nations: Vienna. 2005.

http://www.incb.org/pdf/e/estim/trainmat/NAR\_1%20English%202005.pdf

<sup>50</sup> Anderson, E.D., Davis, C.S. & Burris, S. Achieving Balance Through Locally Feasible Control: A Guide to Applying the Single Convention's Flexible Requirements for the Regulation of Opioids for Medical Purposes. Draft. Drug Control and Access to Medicines (DCAM) Consortium. 2009.

<sup>51</sup> United Nations. *Single Convention on Narcotic Drugs, 1961*. United Nations: New York. 1961.

<sup>52</sup> Anderson, E.D., Davis, C.S. & Burris, S. Achieving Balance Through Locally Feasible Control: A Guide to Applying the Single Convention's Flexible Requirements for the Regulation of Opioids for Medical Purposes. Draft. Drug Control and Access to Medicines (DCAM) Consortium. 2009.

<sup>53</sup> United Nations. *Single Convention on Narcotic Drugs*, 1961. United Nations: New York. 1961.

<sup>54</sup> United Nations. *Single Convention on Narcotic Drugs*, 1961. United Nations: New York. 1961.

<sup>55</sup> United Nations. *Single Convention on Narcotic Drugs*, 1961. United Nations: New York. 1961.

<sup>56</sup> United Nations. *Single Convention on Narcotic Drugs*, 1961. United Nations: New York. 1961.

<sup>57</sup> Anderson, E., Beletsky, L., Burris, S., Davis, C. & Kresina, T. (editors) *Closing the Gap: Case Studies of Opioid Access Reform in China, India, Romania and Vietnam.* Legal Studies Research Paper Series. No. 2009-25. Temple University, Beasley School of Law, Center for Health Law, Policy and Practice: Philadelphia. 2009.

<sup>58</sup> Mosoiu, D., Ryan, K.M., Joranson, D.E. & Garthwaite, J.P. 'Reforming drug control policy for palliative care in Romania'. *Lancet* 367: 2110-2117. 2006. http://www.painpolicy.wisc.edu/publicat/06lancet/lancet06.pdf

<sup>59</sup> Mosoiu, D., Mungiu, O.C., Gigore, B. & Landon, A. 'Romania: Changing the regulatory environment'. *J Pain Symptom Manage* 33: 610-614. 2007.

<sup>60</sup> Anderson, E.D., Davis, C.S. & Burris, S. Achieving Balance Through Locally Feasible Control: A Guide to Applying the Single Convention's Flexible Requirements for the Regulation of Opioids for Medical Purposes. Draft. Drug Control and Access to Medicines (DCAM) Consortium. 2009.

<sup>61</sup> Poppy Advisory & Control Board. *The Industry*. Website of the Tasmanian Department of Justice: Hobart. www.justice.tas.gov.au/poppy/the\_industry – accessed 10 March 2010.

<sup>62</sup> Poppy Advisory & Control Board. *Poppy Advisory & Control Board: About us*. Website of the Tasmanian Department of Justice: Hobart. http://www.justice.tas.gov.au/poppy/about\_us – accessed 03 January 2010.

<sup>63</sup> Poppy Advisory & Control Board. *Becoming a grower: Security issues*. Website of the Tasmanian Department of Justice: Hobart. http://www.justice.tas.gov.au/poppy/becoming\_a\_grower/security\_issues – accessed 03 January 2010.

<sup>64</sup> Tedmanson, S. 'Wallabies get high in Tasmanian poppy fields'. *The Times*, 26 June 2009. http://www.timesonline.co.uk/tol/news/world/article6573582.ece

<sup>65</sup> Poppy Advisory & Control Board. *Dangers of poppies*. Website of the Tasmanian Department of Justice: Hobart. http://www.justice.tas.gov.au/poppy/dangers\_of\_poppies – accessed 03 January 2010.

<sup>66</sup> Anderson, E.D., Davis, C.S. & Burris, S. Achieving Balance Through Locally Feasible Control: A Guide to Applying the Single Convention's Flexible Requirements for the Regulation of Opioids for Medical Purposes. Draft. Drug Control and Access to Medicines (DCAM) Consortium. 2009.

<sup>67</sup> United Kingdom Home Office. *Guidelines for the Safe Custody of Controlled Drugs in Transit*. UK Home Office: London. 2003.

<sup>68</sup> Anderson, E.D., Davis, C.S. & Burris, S. Achieving Balance Through Locally Feasible Control: A Guide to Applying the Single Convention's Flexible Requirements for the Regulation of Opioids for Medical Purposes. Draft. Drug Control and Access to Medicines (DCAM) Consortium. 2009.

<sup>69</sup> International Narcotics Control Board (INCB). *Report of the International Narcotics Control Board for 2009*. INCB: New York, NY. 2010. http://www.incb.org/incb/en/annual-report-2009.html

<sup>70</sup> United Nations. *Single Convention on Narcotic Drugs, 1961*. United Nations: New York. 1961.

<sup>71</sup> International Narcotics Control Board (INCB). *Demand for and Supply of Opiates for Medical and Scientific Needs*. United Nations: New York. 1989.

<sup>72</sup> World Health Organization (WHO). Achieving Balance in National Opioids Control Policy: Guidelines for Assessment. WHO: Geneva. 2000.

<sup>73</sup> Cherny, N.I., Baselga, J., de Conno, F. & Radbruch, L. 'Formulary availability and regulatory barriers to accessibility of opioids for cancer pain in Europe: a report from the ESMO/EAPC Opioid Policy Initiative'. *Oxford Journals, Annals of Oncology*. Volume 21, Issue 3. 2010.

<sup>74</sup> Human Rights Watch. *Please, do not make us suffer any more: Access to Pain Treatment as a Human Right.* Human Rights Watch: New York. 2009.

<sup>75</sup> Cherny, N.I., Baselga, J., de Conno, F. & Radbruch, L. 'Formulary availability and regulatory barriers to accessibility of opioids for cancer pain in Europe: a report from the ESMO/EAPC Opioid Policy Initiative'. *Oxford Journals, Annals of Oncology*. Volume 21, Issue 3. 2010.

<sup>76</sup> World Health Organization (WHO). *Cancer Pain Relief.* Second Edition. 1996.

<sup>77</sup> Degenhardt, L., Larance, B., Mathers, B., Azim, T., Kamarulzaman, A., Mattick, R., Panda, S., Toufik, A., Tyndall, M., Wiessing, L. & Wodak, A. *Benefits and risks of pharmaceutical opioids: Essential treatment and diverted medication – A global review of availability, extra-medical use, injection and the association with HIV.* National Drug and Alcohol Research Centre (NDARC), University of New South Wales (UNSW): Sydney. 2007.

<sup>78</sup> Joranson, D.E. & Gilson, A.M. 'A much-needed window on opioid diversion (Editorial)'. *Pain Med* 8(2): 128-129. 2007. http://www.painpolicy.wisc.edu/publicat/07painmed/07painmed.pdf

<sup>79</sup> Degenhardt, L., Larance, B., Mathers, B., Azim, T., Kamarulzaman, A., Mattick, R., Panda, S., Toufik, A., Tyndall, M., Wiessing, L. & Wodak, A. *Benefits and risks of pharmaceutical opioids: Essential treatment and diverted medication – A global review of availability, extra-medical use, injection and the association with HIV.* National Drug and Alcohol Research Centre (NDARC), University of New South Wales (UNSW): Sydney. 2007.

<sup>80</sup> International Narcotics Control Board (INCB). *Report of the International Narcotics Control Board for 2009*. INCB: New York, NY. 2010. http://www.incb.org/incb/en/annual-report-2009.html

<sup>81</sup> Cherny, N.I., Baselga, J., de Conno, F. & Radbruch, L. 'Formulary availability and regulatory barriers to accessibility of opioids for cancer pain in Europe: a report from the ESMO/EAPC Opioid Policy Initiative'. *Oxford Journals, Annals of Oncology*. Volume 21, Issue 3. 2010.

<sup>82</sup> Goldenbaum, D.M., Christopher, M., Gallagher, R.M., Fishman, S., Payne, R., Joranson, D., Edmondson, D., McKee, J. & Thexton, A.'Physicians Charged with Opioid Analgesic-Prescribing Offenses'. *Pain Medicine*, Vol. 9, No. 6. 2008.

<sup>83</sup> Human Rights Watch. *Please, do not make us suffer any more: Access to Pain Treatment as a Human Right.* Human Rights Watch: New York. 2009. <sup>84</sup> Degenhardt, L., Larance, B., Mathers, B., Azim, T., Kamarulzaman, A., Mattick, R., Panda, S., Toufik, A., Tyndall, M., Wiessing, L. & Wodak, A. *Benefits and risks of pharmaceutical opioids: Essential treatment and diverted medication – A global review of availability, extra-medical use, injection and the association with HIV.* National Drug and Alcohol Research Centre (NDARC), University of New South Wales (UNSW): Sydney. 2007.

<sup>85</sup> Cherny, N.I., Baselga, J., de Conno, F. & Radbruch, L. 'Formulary availability and regulatory barriers to accessibility of opioids for cancer pain in Europe: a report from the ESMO/EAPC Opioid Policy Initiative'. *Oxford Journals, Annals of Oncology*. Volume 21, Issue 3. 2010.

<sup>86</sup> Degenhardt, L., Larance, B., Mathers, B., Azim, T., Kamarulzaman, A., Mattick, R., Panda, S., Toufik, A., Tyndall, M., Wiessing, L. & Wodak, A. *Benefits and risks of pharmaceutical opioids: Essential treatment and diverted medication – A global review of availability, extra-medical use, injection and the association with HIV.* National Drug and Alcohol Research Centre (NDARC), University of New South Wales (UNSW): Sydney. 2007.

<sup>87</sup> Joranson, D.E. & Gilson, A.M. 'Drug crime is a source of abused pain medications in the United States (Letter)'. *J Pain Symptom Manage* 30(4): 299-301. 2005. http://www.painpolicy.wisc.edu/publicat/05jpsm/05jpsm.pdf

<sup>88</sup> Degenhardt, L., Larance, B., Mathers, B., Azim, T., Kamarulzaman, A., Mattick, R., Panda, S., Toufik, A., Tyndall, M., Wiessing, L. & Wodak, A. *Benefits and risks of pharmaceutical opioids: Essential treatment and diverted medication – A global review of availability, extra-medical use, injection and the association with HIV.* National Drug and Alcohol Research Centre (NDARC), University of New South Wales (UNSW): Sydney. 2007.

<sup>89</sup> 2008 Statement from the Alliance of State Pain Initiatives about Prescription Monitoring Programs. http://www.aspi.wisc.edu/documents/pdf/PMPStatement2008.pdf

<sup>90</sup> Joranson, D.E. & Gilson, A.M. 'Drug crime is a source of abused pain medications in the United States (Letter)'. *J Pain Symptom Manage* 30(4): 299-301. 2005. http://www.painpolicy.wisc.edu/publicat/05jpsm/05jpsm.pdf

<sup>91</sup> Joranson, D.E. & Gilson, A.M. 'Wanted: A public health approach to prescription opioid abuse and diversion (Editorial)'. *Pharmacoepidemiology and Drug Safety* 15: 632-634. 2006. http://www.painpolicy.wisc.edu/publicat/pharmaco/commentary.pdf

<sup>92</sup> Joranson, D.E. & Gilson, A.M. 'A much-needed window on opioid diversion (Editorial)'. *Pain Med* 8(2): 128-129. 2007. http://www.painpolicy.wisc.edu/publicat/07painmed/07painmed.pdf

<sup>93</sup> International Narcotics Control Board (INCB). *Report of the International Narcotics Control Board*. INCB: New York, NY. 2006.

<sup>94</sup> Smith, B., Miller, P., O'Keefe, B. & Fry, C. *Benzodiazepine and pharmaceutical opioid misuse and their relationship to crime. An examination of illicit prescription drug markets in Melbourne, Hobart and Darwin. Victorian Report.* National Drug Strategy. Monograph Series, No. 23. Turning Point Alcohol and Drug Centre: Melbourne. 2007.

<sup>95</sup> Anderson, E.D., Davis, C.S. & Burris, S. Achieving Balance Through Locally Feasible Control: A Guide to Applying the Single Convention's Flexible Requirements for the Regulation of Opioids for Medical Purposes. Draft. Drug Control and Access to Medicines (DCAM) Consortium. 2009.

<sup>96</sup> Burris, S. & Davis, C.S. *Blueprint for Reforming Access to Therapeutic Opioids: Entry Points for International Action to Remove the Policy Barriers to Care.* Centers for Law and the Public's Health: A Collaborative at the Johns Hopkins and Georgetown Universities. 2008.

<sup>97</sup> International Harm Reduction Development (IHRD). *Harm Reduction Developments*. IHRD Program of the Open Society Institute: New York. 2008.

<sup>98</sup> International Harm Reduction Development (IHRD). *Barriers to Access: Medication-Assisted Treatment and Injection-Driven HIV Epidemics*. IHRD Program of the Open Society Institute: New York. 2008. http://www.soros.org/initiatives/health/focus/ihrd/articles\_publications/publications/barriers\_20080215/barriersfootn otes040808.pdf – accessed 22 January 2010.

<sup>99</sup> International Harm Reduction Development (IHRD). *Harm Reduction Developments*. IHRD Program of the Open Society Institute: New York. 2008.

<sup>100</sup> International Harm Reduction Development (IHRD). *Barriers to Access: Medication-Assisted Treatment and Injection-Driven HIV Epidemics*. IHRD Program of the Open Society Institute: New York. 2008. http://www.soros.org/initiatives/health/focus/ihrd/articles\_publications/publications/barriers\_20080215/barriersfootn otes040808.pdf – accessed 22 January 2010.

<sup>101</sup> International Harm Reduction Development (IHRD). *Harm Reduction Developments*. IHRD Program of the Open Society Institute: New York. 2008.

<sup>102</sup> International Harm Reduction Development (IHRD). *Harm Reduction Developments*. IHRD Program of the Open Society Institute: New York. 2008.

<sup>103</sup> International Harm Reduction Development (IHRD). *Harm Reduction Developments*. IHRD Program of the Open Society Institute: New York. 2008.

<sup>104</sup> World Health Organization (WHO). Achieving Balance in National Opioids Control Policy: Guidelines for Assessment. WHO: Geneva. 2000.

<sup>105</sup> United Nations. *Single Convention on Narcotic Drugs*, 1961. United Nations: New York. 1961.

<sup>106</sup> Smith, B., Miller, P., O'Keefe, B. & Fry, C. *Benzodiazepine and pharmaceutical opioid misuse and their relationship to crime. An examination of illicit prescription drug markets in Melbourne, Hobart and Darwin. Victorian Report.* National Drug Strategy. Monograph Series, No. 23. Turning Point Alcohol and Drug Centre: Melbourne. 2007.

<sup>107</sup> Smith, B., Miller, P., O'Keefe, B. & Fry, C. *Benzodiazepine and pharmaceutical opioid misuse and their relationship to crime. An examination of illicit prescription drug markets in Melbourne, Hobart and Darwin. Victorian Report.* National Drug Strategy. Monograph Series, No. 23. Turning Point Alcohol and Drug Centre: Melbourne. 2007.

<sup>108</sup> Burris, S., Blankenship, K., Donoghoe, M. et al. 'Addressing the "Risk Environment" for Injection Drug Users: The Mysterious Case of the Missing Cop'. *The Milbank Quarterly* 82, No. 1:125. 2004.

<sup>109</sup> International Narcotics Control Board (INCB). *Report of the International Narcotics Control Board for 2009*. INCB: New York, NY. 2010. http://www.incb.org/incb/en/annual-report-2009.html

<sup>110</sup> Degenhardt, L., Larance, B., Mathers, B., Azim, T., Kamarulzaman, A., Mattick, R., Panda, S., Toufik, A., Tyndall, M., Wiessing, L. & Wodak, A. *Benefits and risks of pharmaceutical opioids: Essential treatment and diverted medication – A global review of availability, extra-medical use, injection and the association with HIV.* National Drug and Alcohol Research Centre (NDARC), University of New South Wales (UNSW): Sydney. 2007.

<sup>111</sup> Human Rights Watch. *Please, do not make us suffer any more: Access to Pain Treatment as a Human Right.* Human Rights Watch: New York. 2009.

<sup>112</sup> 2008 Statement from the Alliance of State Pain Initiatives about Prescription Monitoring Programs. http://www.aspi.wisc.edu/documents/pdf/PMPStatement2008.pdf <sup>113</sup> Burris, S. & Davis, C.S. *Blueprint for Reforming Access to Therapeutic Opioids: Entry Points for International Action to Remove the Policy Barriers to Care.* Centers for Law and the Public's Health: A Collaborative at the Johns Hopkins and Georgetown Universities. 2008.

<sup>114</sup> Anderson, E., Beletsky, L., Burris, S., Davis, C. & Kresina, T. (editors) *Closing the Gap: Case Studies of Opioid Access Reform in China, India, Romania and Vietnam.* Legal Studies Research Paper Series. No. 2009-25. Temple University, Beasley School of Law, Center for Health Law, Policy and Practice: Philadelphia. 2009.

<sup>115</sup> Anderson, E., Beletsky, L., Burris, S., Davis, C. & Kresina, T. (editors) *Closing the Gap: Case Studies of Opioid Access Reform in China, India, Romania and Vietnam.* Legal Studies Research Paper Series. No. 2009-25. Temple University, Beasley School of Law, Center for Health Law, Policy and Practice: Philadelphia. 2009.

<sup>116</sup> Yanjun, S., Changli, W., Ling, W., Ai-lian Woo, J.C., Sabrina, K., Chang, L. & Lei, Z. A survey on physician knowledge and attitudes towards clinical use of morphine for cancer pain treatment in China. Support Care Cancer. 2009.

<sup>117</sup> Yanjun, S., Changli, W., Ling, W., Ai-lian Woo, J.C., Sabrina, K., Chang, L. & Lei, Z. A survey on physician knowledge and attitudes towards clinical use of morphine for cancer pain treatment in China. Support Care Cancer. 2009.

<sup>118</sup> Anderson, E., Beletsky, L., Burris, S., Davis, C. & Kresina, T. (editors) *Closing the Gap: Case Studies of Opioid Access Reform in China, India, Romania and Vietnam.* Legal Studies Research Paper Series. No. 2009-25. Temple University, Beasley School of Law, Center for Health Law, Policy and Practice: Philadelphia. 2009.

<sup>119</sup> Joranson, D.E., Cai, Z.J. & Gilson, A.M. 'Barriers to opioid availability in China'. *Bulletin on Drug Dependence* 4(2). 1995.

<sup>120</sup> Zhang, H., Gu, W., Joranson, D.E. & Cleeland, C.S. 'People's Republic of China: Status of cancer pain and palliative care'. *Journal of Pain and Symptom Management* 12(2): 124-126. 1996.

<sup>121</sup> Mazerolle, L., Soole, D. & Rombouts, S. 'Drug Law Enforcement. A Review of the Evaluation Literature'. *Police Quarterly*, Vol. 10, No. 2, 115-153. 2007.

<sup>122</sup> Caulkins, J.P. & Heinz, H.J. *Law Enforcement's Role in a Harm Reduction Regime*. Crime and Justice Bulliten No 64. NSW Bureau of Crime Statistics and Research: Sydney. 2002.

<sup>123</sup> New South Wales Police. *Methadone and other Pharmacotherapies: Guidelines for Police*. NSW Police: Sydney. 2003.

 $^{124}$  Ganguly, J. A legally balanced approach to HIV/AIDS prevention among drug users. Management of drugs and other related issues by Law Enforcements (Police) in Kolkata, India – a unique and best practice model. In *Law Enforcement and Harm Reduction*. 2010.

<sup>125</sup> Reid, G. *Responding to the drug using situation in Calcutta: embracing the holistic approach*. Macfarlane Burnet Centre for Medical Research: Melbourne. 2001.

<sup>126</sup> HIV/AIDS Asia Regional Program (HAARP). *Law Enforcement and Harm Reduction: Advocacy and Action Rationale*. http://www.haarp-online.org – accessed 03 January 2010.

<sup>127</sup> Spooner, C., McPherson, M., & Hall, W. *The role of police in preventing and minimising illicit drug use and its harms*. National Drug Law Enforcement Research Fund Monograph Series No. 2. Commonwealth of Australia: Canberra. 2004.

<sup>128</sup> AIDS Foundation East West (AFEW). *Harm reduction: theory and practice*. Leaflet for militia (unofficial translation). AFEW: Kiev. 2001.

<sup>129</sup> New South Wales Police. *Methadone and other Pharmacotherapies: Guidelines for Police*. NSW Police: Sydney. 2003.

<sup>130</sup> International Narcotics Control Board (INCB). *The principle of proportionality and drug-related offences*. Annual Report. 2007. http://www.incb.org/pdf/annual-report/2007/en/chapter-01.pdf

<sup>131</sup> Reid, G. *Responding to the drug using situation in Calcutta: embracing the holistic approach*. Macfarlane Burnet Centre for Medical Research: Melbourne. 2001.

<sup>132</sup> Ganguly, J. A legally balanced approach to HIV/AIDS prevention among drug users. Management of drugs and other related issues by Law Enforcements (Police) in Kolkata, India – a unique and best practice model. In *Law Enforcement and Harm Reduction*. 2010.

<sup>133</sup> HIV/AIDS Asia Regional Program (HAARP). *Law Enforcement and Harm Reduction Curriculum*. http://www.haarp-online.org/www/html/7-home-page.asp

<sup>134</sup> HIV/AIDS Asia Regional Program (HAARP). *Law Enforcement and Harm Reduction Curriculum*. http://www.haarp-online.org/www/html/7-home-page.asp

<sup>135</sup> Drug and Alcohol Strategy Unit. *About Victoria Police*. 2007.
http://www.police.vic.gov.au/content.asp?Document\_ID=5059 – accessed 20 March 2010

<sup>136</sup> Cannabis Cautioning Scheme in New South Wales. *Summary*. http://www.druginfo.nsw.gov.au/diversion/cannabis\_cautioning\_scheme/cannabis\_cautioning\_scheme\_in\_nsw\_sum mary – accessed 20 March 2010.

<sup>137</sup> United Kingdom Drug Policy Commission. *Refocusing Drug-Related Law Enforcement to Address Harms*. Full review report. UK Drug Policy Commission: London. 2009.

<sup>138</sup> United Kingdom Drug Policy Commission. *Refocusing Drug-Related Law Enforcement to Address Harms*. Full review report. UK Drug Policy Commission: London. 2009.

<sup>139</sup> King, R.S. & Pasquarella, J. Drug Courts: A Review of the Evidence. The Sentencing Project. 2009.

<sup>140</sup> King, R.S. & Pasquarella, J. Drug Courts: A Review of the Evidence. The Sentencing Project. 2009.

<sup>141</sup> United States Department of Justice. Adult Drug Court Survey Report: Program Operations, Services and Participant Perspectives: Part 5 - Law Enforcement Agency Perspectives. OJP Drug Court Clearinghouse and Technical Assistance Project. A Project of the Drug Courts Program Office, Office of Justice Programs, U.S. Department of Justice. 2001.

<sup>142</sup> United States Department of Justice. Adult Drug Court Survey Report: Program Operations, Services and Participant Perspectives: Part 5 - Law Enforcement Agency Perspectives. OJP Drug Court Clearinghouse and Technical Assistance Project. A Project of the Drug Courts Program Office, Office of Justice Programs, U.S. Department of Justice. 2001.

<sup>143</sup> About Victoria Police. Alcohol and Drug Strategy Unit.
http://www.police.vic.gov.au/content.asp?Document\_ID=5059 (accessed March 20, 2010)

<sup>144</sup> National Drug Court Institute. *Quality Improvement for Drug Courts: Evidence-Based Practices*. National Drug Court Institute: Alexandria, Virginia. 2008.

<sup>145</sup> King, R.S. & Pasquarella, J. Drug Courts: A Review of the Evidence. The Sentencing Project. 2009.

<sup>146</sup> National Drug Court Institute. *Quality Improvement for Drug Courts: Evidence-Based Practices*. National Drug Court Institute: Alexandria, Virginia. 2008.

<sup>147</sup> Drug and Alcohol Strategy Unit. *About Victoria Police*. 2007.
http://www.police.vic.gov.au/content.asp?Document\_ID=5059 – accessed 20 March 2010.

<sup>148</sup> HIV/AIDS Asia Regional Program (HAARP). *Law Enforcement and Harm Reduction Curriculum*. http://www.haarp-online.org/www/html/7-home-page.asp

<sup>149</sup> Dolan, K., Khoei, E.M., Brentar, C. & Stevens, A. *Prisons and Drugs: A global review of incarceration, drug use and drug services*. The Beckley Foundation Drug Policy Programme. Report 12. 2007.

<sup>150</sup> International Harm Reduction Development (IHRD). *Harm Reduction Developments*. IHRD Program of the Open Society Institute: New York. 2008.

<sup>151</sup> Ali, R., Burrows, D., Gowing, L., Vial, R., Walsh, N., Onyancha, P. & Finnerty, E. Architecture and Country Level Practice on HIV/AIDS and Drug Treatment. AIDS Project Management Group (APMG): Sydney. 2008.

<sup>152</sup> Dolan, K., Khoei, E.M., Brentar, C. & Stevens, A. *Prisons and Drugs: A global review of incarceration, drug use and drug services*. The Beckley Foundation Drug Policy Programme. Report 12. 2007.

<sup>153</sup> Queensland Corrective Services (QCS). *Standard Guidelines for Corrections in Australia*. 2004: 20. http://www.correctiveservices.qld.gov.au/Resources/Policies/Documents/guidelines\_for\_corrections\_in\_aus.pdf – accessed 03 January 2010.

<sup>154</sup> World Health Organization (WHO). Guidelines on HIV Infection and AIDS in Prison. WHO: Geneva. 2007.

<sup>155</sup> World Health Organization (WHO). *Interventions to Address HIV in Prisons: Drug Dependence Treatments*. Evidence for Action Technical Papers. WHO: Geneva. 2007.

<sup>156</sup> World Health Organization (WHO). *Interventions to Address HIV in Prisons: Drug Dependence Treatments*. Evidence for Action Technical Papers. WHO: Geneva. 2007.

<sup>157</sup> Grund, J. Outbreak of HIV Infection in the Lithuanian Prison System Assessment of the Situation and the Developing Response. UNAIDS: Geneva. 2002.

<sup>158</sup> Thaisri, H. 'HIV infection and risk factors among Bangkok prisoners, Thailand: a prospective cohort study'. *BMC Infectious Diseases*, 3:45. 2003.

<sup>159</sup> Grund, J. Outbreak of HIV Infection in the Lithuanian Prison System Assessment of the Situation and the Developing Response. UNAIDS: Geneva. 2002.

<sup>160</sup> World Health Organisation (WHO). Evidence for Action: *Reduction of HIV Transmission in Prisons*. WHO: Geneva. 2004.

<sup>161</sup> American Association for the Treatment of Opioid Dependence (AATOD). *Fact Sheet for Drug Courts: Methadone Maintenance and other Pharmacotherapeutic Interventions in the Treatment of Opioid Dependence.* AATOD: New York. 2004. http://www.aatod.org/fact\_drug\_court.html

<sup>162</sup> WHO/UNODC. *Principles of Drug Dependence Treatment*. 2008. http://www.who.int/substance\_abuse/publications/principles\_drug\_dependence\_treatment.pdf

<sup>163</sup> World Health Organization (WHO). *Interventions to address HIV in prisons: HIV care, treatment and support*. Evidence for Action Technical Papers. WHO: Geneva, 2007.

<sup>164</sup> Ogden, E. 'Showing care in custody'. *Of Substance*. Vol. 8, No. 1. 2010.

<sup>165</sup> World Health Organization (WHO). *Interventions to address HIV in prisons: HIV care, treatment and support*. Evidence for Action Technical Papers. WHO: Geneva, 2007.

<sup>166</sup> American Association for the Treatment of Opioid Dependence (AATOD). *Fact Sheet for Drug Courts: Methadone Maintenance and other Pharmacotherapeutic Interventions in the Treatment of Opioid Dependence.* AATOD: New York. 2004. http://www.aatod.org/fact\_drug\_court.html

<sup>167</sup> Office of the Correctional Services Commissioner (OCSC). *Victorian Prison Opioid Substitution Therapy Program: Clinical and Operational Policy and Procedures*. OCSC: Melbourne. 2003.

<sup>168</sup> World Health Organization (WHO). *Interventions to address HIV in prisons: HIV care, treatment and support*. Evidence for Action Technical Papers. WHO: Geneva, 2007.

<sup>169</sup> World Health Organization (WHO). *Towards universal access: scaling up priority HIV/AIDS interventions in the health sector*. WHO: Geneva. 2007.

<sup>170</sup> Ganguly, J. A legally balanced approach to HIV/AIDS prevention among drug users. Management of drugs and other related issues by Law Enforcements (Police) in Kolkata, India – a unique and best practice model. In Law Enforcement and Harm Reduction. 2010.

<sup>171</sup> International Harm Reduction Development (IHRD). *Barriers to Access: Medication-Assisted Treatment and Injection-Driven HIV Epidemics*. IHRD Program of the Open Society Institute: New York. 2008. http://www.soros.org/initiatives/health/focus/ihrd/articles\_publications/publications/barriers\_20080215/barriersfootn otes040808.pdf – accessed 22 January 2010.

<sup>172</sup> Dolan, K., Khoei, E.M., Brentar, C. & Stevens, A. *Prisons and Drugs: A global review of incarceration, drug use and drug services*. The Beckley Foundation Drug Policy Programme. Report 12. 2007.

<sup>173</sup> American Association for the Treatment of Opioid Dependence (AATOD). *Fact Sheet for Drug Courts: Methadone Maintenance and other Pharmacotherapeutic Interventions in the Treatment of Opioid Dependence.* AATOD: New York. 2004. http://www.aatod.org/fact\_drug\_court.html

<sup>174</sup> Ogden, E. 'Showing care in custody'. *Of Substance*. Vol. 8, No. 1. 2010.

<sup>175</sup> Ogden, E. 'Showing care in custody'. *Of Substance*. Vol. 8, No. 1. 2010.

<sup>176</sup> Office of the Correctional Services Commissioner (OCSC). *Victorian Prison Opioid Substitution Therapy Program: Clinical and Operational Policy and Procedures*. OCSC: Melbourne. 2003.

<sup>177</sup> Queensland Corrective Services (QCS). *Drug Strategy 2006: Tackling Drug Abuse and Addiction Changing Lives in Queensland Prisons*. QCS: Brisbane. 2006: 24.

<sup>178</sup> HIV/AIDS Asia Regional Program (HAARP). *Law Enforcement and Harm Reduction Curriculum*. http://www.haarp-online.org/www/html/7-home-page.asp – accessed 03 January 2010.

<sup>179</sup> HIV/AIDS Asia Regional Program (HAARP). *Law Enforcement and Harm Reduction Curriculum*. http://www.haarp-online.org/www/html/7-home-page.asp – accessed 03 January 2010.

<sup>180</sup> Markman, M. Care of the Incarcerated Cancer Patient. University of Texas. 2007.

<sup>181</sup> Lin, J.T. & Mathew, P. 'Cancer Pain Management in Prisons: A Survey of Primary Care Practitioners and Inmates'. *Journal of Pain and Symptom Management*. Volume 29, Issue 5: 466-473. 2005.

<sup>182</sup> World Health Organization (WHO). *Interventions to Address HIV in Prisons: Drug Dependence Treatments*. Evidence for Action Technical Papers. WHO: Geneva. 2007.

<sup>183</sup> World Health Organization (WHO). *Cancer Pain Relief.* Second Edition. 1996.

<sup>184</sup> Human Rights Watch. *Please, do not make us suffer any more: Access to Pain Treatment as a Human Right.* Human Rights Watch: New York. 2009.

<sup>185</sup> International Narcotics Control Board (INCB). *Report of the International Narcotics Control Board for 2009*. INCB: New York, NY. 2010. http://www.incb.org/incb/en/annual-report-2009.html